

Dynamic monopoly and consumers profiling accuracy

Didier Laussel, Ngo Van Long, Joana Resende

▶ To cite this version:

Didier Laussel, Ngo Van Long, Joana Resende. Dynamic monopoly and consumers profiling accuracy. Journal of Economics and Management Strategy, 2022, 31 (3), pp.579-608. 10.1111/jems.12479 . hal-03665780

HAL Id: hal-03665780 https://amu.hal.science/hal-03665780v1

Submitted on 27 May 2022 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Dynamic monopoly and consumers profiling accuracy

Didier Laussel¹

Ngo Van Long^{2,3} D Joana Resende⁴

¹CNRS & EHESS, Aix-Marseille University (Aix-Marseille School of Economics), Marseille, France ²Department of Economics, McGill University, Montreal, Quebec, Canada ³Hitotsubashi Institute for Advanced Study, Hitotsubashi University, Tokyo, Japan ⁴U.Porto-CEF.UP, Porto, Portugal

Abstract

Using a Markov-perfect equilibrium model, we show that the use of customer data to practice intertemporal price discrimination will improve monopoly profit if and only if information precision is higher than a certain threshold level. This U-shaped relationship lends support to a popular view that knowledge is good only if it is sufficiently refined. When information accuracy can only be achieved through costly investment, we find that investing in profiling is profitable only if this allows to reach a high enough level of information precision. Consumers expected surplus being a hump-shaped function of information accuracy, we show that consumers have an incentive to lobby for privacy protection legislation which raises the cost of monopoly's investment in information accuracy. However, this cost should not dissuade firms to collect some information on customers' tastes, as the absence of consumers' profiling is actually detrimental to consumers.

KEYWORDS

big data, consumers' collective action on privacy protection legislation, consumers profiling, dynamic monopoly, endogenous investment in profiling capability

INTRODUCTION 1

Recent advances in digital technologies have enabled firms to collect, store and process "big data" to gain information on customers and thus enhance their profits by treating different types of customers differently.¹ The collection of huge amounts of consumer-specific information together with the use of complex algorithms may partly or even totally reveal individual customers' preferences, allowing firms to engage in increasingly complex price discrimination strategies. As an illustrative instance of this reality, we may cite Uber's route-based pricing which tries to implement a new fare system that charges customers what it predicts they are willing to pay.²

Although in some cases, firms will only get coarse information on customers' characteristics³ (for instance, the date when she first bought the good or service), in other cases firms have access to an impressive set of information⁴, which may include customers' web browsing history, demographics, real time location, online purchasing/consumption habits, social network activities and connections, and son on... as illustrated with the Uber example above, sometimes information becomes so precise that firms are able to get precise estimates of consumers' willingness to pay (WTP) for the commodity.5

As a matter of fact, the information that firms are able to collect on consumers may be more or less fine depending on a wide variety of factors, such as (i) the state of the information technologies, (ii) the amount the firms invest in their consumers' profiling capability (e.g., buying the hardware and software and/or hiring data analysts), and (iii) the state of legislation on consumers data protection (such as the EU General Data Protection Regulation (GDPR) of 2016

or the California Consumer Privacy Act of 2018), which may itself be influenced by consumers organizations' lobbying activities.

In all cases, taking advantage of a more or less perfect "consumer addressability" (Blattberg & Deighton, 1991), firms may be able to engage in sophisticated price discrimination/market segmentation strategies. More precisely, firms may target pricing, marketing and even product characteristics to groups of consumers or even to individual ones. In this paper, we identify conditions under which firms and consumers may benefit or, on the contrary, are hurt by the firms' capability to collect information on their customers and by the precision of such information. These variables critically determine the scope of firms' market segmentation strategies, which may vary from no segmentation at all (when firms do not collect any information on their customers) to hyper market segmentation (where each individual customer emerges as an independent market segment, allowing for personalized interaction with the firm). In between, we may find a wide diversity of cases (e.g., in some circumstances firms are only able to recognize the consumers' identity without getting any specific information on their preferences; in other cases firms may be able to reckon the preferences of a subset of customers only, etc.). For the sake of tractability, the degree of precision of information is defined here, conditional on the fact that a given customer's exact WTP has not been uncovered yet, by the probability that it will be discovered over the next short time interval.⁶

Identifying the conditions under which firms and consumers may benefit (or be hurt) by their capability to gather and make use of increasingly accurate customer information is a prerequisite to the analysis of the profitability of firms investment in profiling and of the motivation and effects of consumers organizations' lobbying activities to influence the scope of data protection policies. These two questions will also be investigated in this paper within our game theoretic framework.

The key features of our model are as follows. A monopolist produces a nondurable homogenous good (or service) which must be consumed instantaneously. Consumers are heterogeneous in terms of their WTPs for the good, which are initially private information. The monopolist makes pricing decisions at discrete points in time. We refer to the time interval between two consecutive price offers as the contractual commitment period. As in Laussel et al. (2020a), we assume that the present monopolist cannot prevent his future selves from exploiting the customers information that they will then have at hand.

The timing of the game we study is as follows. In the first stage of the game, the consumers collectively lobby to influence the toughness of privacy protection legislation to undermine the efficiency of firms' investment in profiling. In the second stage, the monopolist considers whether to invest or not in its big data capability, and, if yes, solves for the optimal amount it invests in profiling accuracy, thus endogenously determining the degree of information precision. The third stage of the game is an infinite-horizon dynamic game between the (partially) informed monopolist and the consumers, in which the former chooses in each period the prices offered to new and old customers while consumers decide, based on their type and their rational expectations of future prices, when to make their first purchase.

We solve the game by backwards induction. Thus, we first demonstrate that, in stage three, monopoly profit is a U-shaped function of the degree of information precision while the expected consumer's surplus is a hump-shaped function of it. These shapes follow from two conflicting effects of consumers profiling on firms' profits. On the one hand, there is a positive direct effect since, other things equal (i.e., keeping unchanged the firm's future strategies), intra-period (finer) price discrimination allowed by more accurate consumers' profiling would increase profits. On the other hand, a negative strategic effect arises as having more information on customers leads to lower introductory prices (targeted to new customers) and lower profits. Such downward pressure on firms' introductory prices tends to be beneficial to the consumers. However, increased precision of firm's information on consumer preferences also enables firms to extract more consumer surplus in future purchases, leading to the consumers' detrimental effect described above. This allows us to make the point that an increase in the precision of firms' customers profiling has ambiguous profit and consumer surplus effects and therefore consumers (firms) may not necessarily gain (lose) from tighter data protection regulation.

Indeed, when we look at the second-stage of our game, we find that the firm's investment must exceed a critical level amount in order for information collection to be worthwhile. This implies that a monopolist should never spend just enough money to obtain very coarse information on rational customers. Moreover, it may not pay off to invest in fully accurate customer profiling technologies, with firms optimally choosing not to disclose the WTP of all customers when this becomes excessively costly.

Finally, when looking at the first stage of the game, we show that, from the consumers' perspectives, the best protective legislation always should reduce firms' investment in profiling but should not always preclude it. When

consumers are able to influence firms' consumer profiling ability (e.g., through tighter or more relaxed data protection rules) and they strategically anticipate firms' optimal investment decisions in the subsequent stage, consumers tend to favor milder data protection regulation: on the one hand, the total prohibition of data collection would actually hurt them (as it would eliminate pricing pressure from the monopolist's future selves); on the other hand, it is in their interest to keep a sufficiently large fraction of returning customers untraceable (to avoid the detrimental effects associated with the use of such information to engage in first-degree price discrimination over returning customers).

To clarify these effects, let us elaborate on the third stage interaction between the firm and the customers, taking as given the degree of precision of customer profiling. Given that the monopolist has chosen to collect and process customer information, in each of the future periods he classifies returning customers according to their first period of purchase ("their vintage"). Moreover, given any vintage *n* of former customers, the monopolist can identify at each point of time the exact WTP of *a fraction* of them. The remaining fraction corresponds to untraceable customers (in the sense that the firm recognizes them as returning consumers but has no information about their preferences). The fraction of identified customers depends on the degree of accuracy of the monopolist's customer profile. Moreover, for any given vintage, the size of the unidentified fraction gets smaller and smaller over time, thanks to the firm's perseverance in collecting and processing data.⁷ Accordingly, the monopolist is able to engage in different types of price discrimination among returning customers. Those include: (i) third-degree price discrimination with respect to the former untraceable customers, and (ii) first-degree price discrimination with respect to "identified customers" (the ones whose exact WTP have been discovered). At the beginning of any given period *n*, the firm faces *n* groups of former customers and decides on the size of the current period's new customers (i.e., the new market segment, denoted as the (n + 1)th market segment) by offering them an attractive "introductory offer" price.

The size of the new market segment is determined by the attractiveness of the new introductory price, which influences the decision of each potential new customer on whether to make her first purchase in that period, or delay it until the next period. Her decision depends on her expectations regarding next period's introductory price and the price she would have to face in the future as a former customer. The firm is fully aware of this. Their strategic interactions lead to the dynamic game played between firm and customers in the third stage. We characterize the Markov Perfect Equilibrium (MPE) of this stage game for any given value of the information precision parameters (which is endogenously chosen in previous stages). The monopolist's equilibrium profit turns out to be U-shaped in the degree of precision of his information about former customers. We show that if the firm's precision of information is high enough such that the rate of decline of the unidentified fraction exceeds a threshold level, its profit will be higher than what is obtained under the no information benchmark (where the static equilibrium is repeated indefinitely and therefore the monopolist avoids profit losses due to price competition from his future selves). Beyond this threshold, greater precision leads to higher profit. At the same time, we also find that the aggregate consumers surplus is a non-monotone, hump-shaped, function of the degree of precision of the firm's information about former customers, evolving in a way opposite to profits.⁸

Our analysis also unveils that whereas it is true that eventually the whole market is covered (in sharp contrast with the static equilibrium outcome), there is an overall positive Coasian correlation between the length of the commitment period and aggregate profit only when the firm's precision of information is very small, the reverse occurring when this precision is very high. Intuitively, a shortening of the commitment interval generates two opposite effects, whose relative importance depends on the degree of precision of the information. The first one is the well-known Coasian mechanism: the potentially new consumers, knowing that the introductory price will go down very quickly, have an incentive to delay their first purchase, triggering an even quicker price decrease for new customers. The second, which goes the other way, is a "front loading effect"⁹: when the commitment period is shorter, the monopolist is able to extract sooner the surplus of those former customers whose exact WTP are discovered. The second effect globally dominates the first one if the rate of decline of the fraction of untraceable returning customers is sufficiently high, namely when it is greater than the rate of interest. The reverse occurs, and the dynamics are globally Coasian, in the opposite case. Looking at the limit value of the aggregate profit when the length of the commitment period becomes infinitesimal, we show that it is equal to zero if and only if the monopolist's information is the coarsest one and that, more generally, in this limiting case, aggregate profit is increasing in the degree of precision of the information.

Having understood the main findings concerning the stage-three equilibrium, the results for stage two and stage one of the game are intuitively plausible. In the second stage, the firm chooses, given the state of the legislation on consumers data protection, whether to invest or not in profiling and, when investing, the amount to invest (or, equivalently, the degree of accuracy of information the firm is able to collect and manipulate to engage in intratemporal price discrimination). The outcome follows from the shape of the (third stage) profit function with respect to the monopolist's level of information and from the parameters of the investment function. Were information collection costless, the firm would always choose to learn perfectly the preferences of its old customers. With costly investment in information collection, it never chooses an investment level that would result only in a "coarse" information level, since not investing at all would then be more profitable. In the very first stage of the game, what is the best protective legislation from the consumers' point of view? This follows from the shape of the consumers surplus function with respect to the degree of information precision (which is optimally chosen by the firm in the second stage). The unconstrained maximum for CS obtains when the monopolist invests just enough to collect the coarsest possible information. This unconstrained maximum is however unattainable since the firm would always avoid that level of investment (and strategic consumers rationally anticipate this outcome). We show that the best legislation from the coarsest information preclude investments in profiling: when the cost of acquiring the coarsest information is low (which is a rather likely scenario), the monopolist will collect information beyond the coarsest level, but a less precise one than without any legislation.

The rest of the paper is organized as follows: Section 2 presents an overview of the related literature. Section 3 introduces the model and the timing of the game, which is solved by backwards induction. Section 4 analyzes the third-stage of the game, investigating firm's price discrimination strategies for a given information precision degree. Section 5 characterizes the MPE of the dynamic game played in the third-stage. Section 6 presents the profit and consumers surplus effects of exogenous changes in firms' profiling ability. Section 7 endogenizes firms' decisions on profiling accuracy by looking at the monopolist's optimal investment decision in the second-stage of the game. Section 8 deals with the first stage of the game, allowing consumers to strategically influence firms' optimal profiling decisions. Finally, Section 9 concludes.

2 RELATED LITERATURE

There is a stream of literature, developed over the past two or three decades, that attempts to assess whether, in the context of oligopolistic competition, better information on customers leads to higher profits for firms. The answer is not obvious, because while better information enables firms to practice price discrimination among their customers, competition among firms becomes fiercer.¹⁰ The early papers focus on the case of a duopoly where information acquisition is exogenous. Corts (1998) for instance focused on the comparison between no discrimination and third-degree price discrimination, without assuming the Hotelling spatial structure. He found that, even under best response asymmetry¹¹, either both prices decrease ("all-out competition") or they both increase ("all-out price gouging") under price discrimination, depending on the relative importance of the two markets, and thus profits may increase or decrease. Thisse and Vives (1988) compared, within the Hotelling spatial model, the outcomes under no discrimination ("mill prices") and under first-degree price discrimination ("delivered prices"), concluding that equilibrium profits are smaller under the latter.¹² Liu and Serfes (2004) also used the Hotelling framework with exogenous information acquisition, but made the quality of information a continuous variable, with two-groups and perfect discrimination being the two extreme situations. They concluded that "equilibrium profits exhibit a U-shaped relationship with respect to information quality" (p. 671).

A more recent strand of literature has considered endogenous information acquisition when firms are able to learn about consumers' purchase history and/or their personal tastes. This behavior-based price discrimination literature (e.g., Chen, 1997; Choe et al., 2018; Fudenberg & Tirole, 2000)¹³ shares with our model the feature that firms learn about consumers' preferences by accessing data on their purchase history and use this information to practice price discrimination (see Fudenberg & Villas-Boas, 2006; Stole, 2007, for surveys). Following the seminal works by Fudenberg and Tirole (2000) and Chen (1997), a number of authors have investigated the competitive effects of behavior-based price discrimination. See, for example, Chen and Pearcy (2010), Esteves (2009, 2010), Gehrig et al. (2011, 2012), and Villas-Boas (1999), just to mention a few. Most of these articles focus on duopolistic competition in a two-period setting and, with some exceptions, they tend to conclude that behavior-based pricing heightens competition and is detrimental to profits. Some of the two-period models, for example, Acquisti and Varian (2005, p. 7), allow new cohort of consumers in period 2. There are also models with an OLG structure, for example, Villas-Boas (1999). Our modeling framework excludes learning within an OLG structure.

Compared with the above literature, our model has a number of novel features. First, instead of considering competition among contemporaneous oligopolists, we consider a current information-seeking monopolist who cannot prevent his future selves from offering a sequence of lower and lower introductory prices, which makes it difficult for

him to induce potential consumers to make their first purchases in the current period. Similar to the oligopoly case, the question is whether the negative competition effect outweighs or does not outweigh the positive direct effect of price targeting but here it is the competition between the current and the future incarnations of the monopolist that reduces his market power. This is reminiscent of the literature on durable good monopoly, starting from Coase (1972), according to which a monopolist who is unable to fully commit to future prices would lose all his monopoly power if the time interval between two consecutive price offers goes to zero.¹⁴ The intuition behind this result is that some customers would delay their purchase rather than accepting the durable-good monopolist's current price because they rationally anticipate a future price decrease. Unlike the Coasian durable-good model where there is no role for acquisition of information on former customers, Laussel et al. (2020a) showed that a nondurable-good monopolist that acquires such information will in the future make use of it to exercise price discrimination between returning former customers of different vintages. In this setting, they found that if the information is very coarse (in particular, their baseline model focuses on the case in which the monopolist's information pertains only to customers' first date of purchase) then a similar Coasian mechanism is at work and the monopolist's aggregate profit is eroded, vanishing in the limit when length of the commitment period goes to zero.¹⁵ In this paper, we allow the degree of precision of information to take any value between zero and infinity. We find that the adverse effect of information on profit no longer holds when the monopolist's precision of information on former customers exceeds a critical value. Beyond this threshold, profit is increasing in the quality of information.¹⁶ The paper also enriches the extant literature on price discrimination based on customers' recognition by endogenizing firms' incentives to make up-front investments to improve the accuracy of their customer profiling technologies and by analyzing the effects of allowing for consumers' collective action on privacy protection legislation.

3 THE MODEL

A monopolist produces a nondurable good (or a service). For simplicity, we set the marginal cost at zero. The firm faces a continuum of infinitely-lived consumers that differ from one another with respect to their WTP, denoted by θ . At each instant, a customer may buy (and instantaneously consume) at most one unit of the good. If a type θ consumer buys a unit of the good at the price *p*, her instantaneous net utility is $\theta - p$. We assume that θ is uniformly distributed over the unit interval, [0, 1]. The fraction of the customer base whose type belongs to any interval [θ_a , θ_b] where $0 \le \theta_a < \theta_b \le 1$ is simply $\theta_b - \theta_a$.

The firm enters the market at time t = 0. It announces its prices (which may be consumer-specific, or groupspecific) at discrete points in time, $\{t_n\}$, where n = 0, 1, 2, ... For all n, the distance between t_{n+1} and t_n is Δ , an exogenous constant. We refer to Δ as the firm's "commitment interval." Over each commitment interval, the prices are unchanged. The interval $[n\Delta, (n + 1)\Delta)$ of the time line is called "period n." The initial period is period zero. In each period, the firm sets targeted prices, that include: (i) an introductory price targeted to new consumers; (ii) a set of personalized prices, targeted to returning traceable customers, whose WTP is known to the firm; and (iii) a set of group prices targeted to cohorts of untraceable customers whose WTP is unknown to the firm (thus it is only able to cluster those customers according to the moment of their first purchase). The relative weight of returning traceable vs untraceable customers depends on the degree of precision of the monopolist's customer profiling. This degree will be an endogenous decision of the monopolist, who needs to do an up-front investment¹⁷ in tracking/profiling technologies before playing the dynamic pricing game with his customers.

More precisely, the timing of the game we study is as follows. In the first stage of the game, the consumers collectively lobby to influence the toughness of privacy protection legislation to undermine the efficiency of firms' investment in profiling, with μ denoting the degree of accuracy of firms' profiling capabilities (later on, we shall provide a more rigorous definition of μ). In the second stage, the monopolist considers whether to invest or not in its big data capability, and, if yes, he solves for the optimal amount he invests in profiling. The third stage of the game is an infinite-horizon dynamic game between the (partially) informed monopolist and the consumers, in which the former chooses in each period the prices offered to new and old customers while consumers decide, based on their type and their rational expectations of future prices, when to make their first purchase.

The game is solved by backwards induction and therefore we start by solving the third-stage of the game, taking μ as given. Customers who purchase the good for the first time in period *i* are called vintage *i* consumers. The set of all customers of vintage *i* is called the market segment *i*. We assume that, thanks to information technology, the monopolist can at least classify former customers according to their vintage. Moreover, making use of its "big data"

capability, as soon as a new customer has made her first purchase, the firm is able to begin the process of trying to uncover the exact WTP of returning customers. However, the process has only limited success since firms can hardly get to know the exact preferences of all their returning customers. To capture this feature, we assume that for any nonnegative integer *i*, at any time $t \ge i$, there is a fraction of former customers of vintage *i* whose exact WTP remains unknown to the firm. Because the firm perseveres in its effort to uncover former customers' WTP, this fraction falls at a constant rate μ per unit of time.¹⁸ In other words, the parameter μ reflects the constraints imposed by the state of information technology and consumers protection legislation (which are to be endogenized in the second and the third stage of the game; more generally, improvements in information technology will increase μ , whereas a tightening of legislations will reduce μ).

Among all individuals who make their first purchases at the beginning of period *i*, by the time period i + 1 begins, the firm has managed to discover the exact WTP of a fraction $(1 - e^{-\mu\Delta})$ of them. In all periods i + j where j = 1, 2, 3, ..., the firm is able to perfectly discriminate these "identified customers" by charging them a personalized price equal to their WTP. This means that those consumers end up with zero surplus. In contrast, in dealing with the fraction $e^{-\mu\Delta}$ of vintage *i* customers whose WTP remain unidentified at the beginning of period i + 1, the firm offers all of them a common returning-customer vintage-specific price, denoted by $p^{\mu}(i, i + 1)$, where *i* refers to the vintage, and i + 1 refers to the period. (The subscript *u* indicates that this price applies only to customers whose exact WTP remains unidentified). More generally, in any period $n \ge i + 1$, a vintage-specific price $p^{\mu}(i, n)$ is offered to the (dwindling) fraction $(e^{-\mu\Delta})^{n-i}$ of vintage *i* customers whose exact WTP remain unidentified.

Given any finite period length $\Delta > 0$, it is useful to define $\delta \equiv e^{-\mu\Delta}$. We refer to δ as "*the degree of coarseness*" of the monopolist's information. The polar case $\delta = 1$ (i.e., $\mu = 0$) corresponds to the scenario where the monopolist is unable to uncover the exact WTP of any of his former customers. We refer to this as "the coarsest information case," because in this case all vintage *i* customers will forever remain tagged as "vintage *i* customers with unidentified WTP." At the opposite extreme, $\delta = 0$ (i.e., μ is infinite) means that the monopolist's information is extremely fine: the monopolist knows the exact WTP of each returning customer (engaging in first-degree price discrimination).

For any n = 0, 1, 2, 3, ..., it will be convenient to denote by $p^u(n, n)$ the "introductory price" offered to new customers in period *n*, that is, those who have never bought the good in earlier periods. A new customer of type θ who buys one unit of the good at each instant of time in this period pays $p^u(n, n)$ per unit at each instant. If we denote by r > 0 the constant instantaneous discount rate, it follows that the customers' resulting discounted stream of net utility for that period is

$$v(\theta, n, n) = (\theta - p^u(n, n)) \int_0^{\Delta} e^{-r\tau} d\tau = (\theta - p^u(n, n)) \frac{1 - \beta}{r},$$

where we have defined $\beta \equiv e^{-r\Delta}$.¹⁹ Concerning former customers belonging to any given vintage i < n, at the beginning of period *n*, the monopolist partitions them into two groups. The first group, which accounts for a fraction $(1 - \delta^{n-i})$ of them, consists of all those whose exact WTP has been discovered. Each of them is charged a personalized price (equal to their WTP), leaving them with zero net utility. The second group, which accounts for the remaining fraction, δ^{n-i} , faces a common price $p^u(i, n)$. We refer to members of this group as "unidentified customers" of vintage *i*. Clearly, the members of this group who have a high enough WTP, $\theta \ge p^u(i, n)$, will purchase the good and obtain $v(\theta, i, n) = (1 - \beta)(\theta - p^u(i, n))/r$, whereas those witk $\Phi^u(i, n)$ will not purchase the good.

4 THIRD-STAGE: PRICE DISCRIMINATION UNDER NONCOMMITMENT

We begin our analysis by focusing on the monopolist's price discrimination among former customers, given that the firm has learnt the exact WTP of some of them while it only knows the date of first purchase of the others. The firm is able to exercise first-degree discrimination to identified former customers, while for unidentified former customers, it is only able to exercise third-degree price discrimination, setting one price for each vintage $p^{u}(i, n)$ is generally different from $p^{u}(j, n)$, for $j \neq i$. After determining these prices, we will consider the monopolist's optimal pricing for new customers. Our main focus is on the noncommitment equilibria of the model, that is; we assume that in any period the monopolist is unable commit to prices it will set for future periods.

4.1 Price discrimination among former customers

Let us begin with the following useful observation. Consumers being rational, if in equilibrium a consumer of type θ' finds it optimal to become a first-time customer in period $i \ge 0$, then it must hold that any consumer of type $\theta'' > \theta'$ also finds it optimal to be a first-time customer in some period i', where $i' \le i$. At the beginning of any period $n \ge 0$, there exist *n* known cut-off types $1 \ge \theta_1 \ge \theta_2 \ge \cdots \ge \theta_n$, and the monopolist makes a decision on market expansion, that is, on the optimal θ_{n+1} .²⁰ As a result, new customers in period *n* are of type $\theta \in (\theta_{n+1}, \theta_n]$. A customer of type θ_{i+1} is called a vintage-*i* marginal customer. By construction, such a customer is indifferent between making her first purchase in period *i* or in period *i* + 1. As mentioned in the previous section, at the beginning of period *n*, the monopolist has discovered the exact WTP of a fraction $1 - \delta^{n-i}$ of vintage *i* customers. As to the remaining fraction δ^{n-i} of them, the only information the monopolist has is that they belong to vintage *i*. At the beginning of period *n*, all consumers whose types belong to $(\theta_n, 1]$ have already purchased the good at least once in previous periods. The monopolist does not offer former customers the introductory price that it proposes to new customers: although the monopolist is not able to identify the WTP of such customers, he is still able to identify their vintage, which translates into the possibility of engaging in third-degree price discrimination among returning customers that belong to different vintages.

In light of this, we have that, in any period *n*, the firm offers to former fully recognizable/identified customers (whose WTP has been identified) a type-dependent personalized price $p^{I}(\theta)$, equal to their WTP, thus extracting their whole surplus, that is, $p^{I}(\theta) = \theta$.²¹ The profits that the monopolist makes in period *n* over the set of identified vintage-*i* former customers equal

$$(1-\delta^{n-i})\int_{\theta_{i+1}}^{\theta_i}\theta d\theta = (1-\delta^{n-i})\left(\frac{\theta_i^2}{2}-\frac{\theta_{i+1}^2}{2}\right).$$

In dealing with vintage *i* former customers whose WTP have not been discovered, the monopolist offers a vintagedependent price, $p^{u}(i, n)$, for each vintage i < n (i = 0, 1, 2, ..., n - 1), so as to maximize the profits it makes from this market segment. Given our assumption that θ is uniformly distributed over the interval [0, 1], the fraction of the customer base whose type belongs to the interval $[\theta_{i+1}, \theta_i]$ where $0 \le \theta_{i+1} < \theta_i \le 1$ is simply $\theta_i - \theta_{i+1}$. It follows that the population size of customers that belong to vintage *i* and whose exact WTP remain unknown to the firm at the beginning of period *n* is $\delta^{n-i}(\theta_i - \theta_{i+1})$. Clearly, offering these customers any $p^{u}(i, n) > \theta_i$ would be a dominated strategy because it would result in zero demand from that market segment. Thus, in period *n*, for these unidentified customers of vintage *i* (where *i* < *n*), the monopolist will offer a price $p^{u}(i, n) \le \theta_i$. Clearly, the quantity $Q_{i,n}$ sold in period *n* to these unidentified customers is given by

$$Q_{i,n} = \begin{cases} \delta^{n-i}(\theta_i - p^u(i, n)) & \text{if } \theta_{i+1} \le p^u(i, n) \le \theta_i, \\ \delta^{n-i}(\theta_i - \theta_{i+1}) & \text{if } 0 \le p^u(i, n) \le \theta_{i+1}. \end{cases}$$

Let $\pi(i, n)$ denote the profit function from sales in period *n* to vintage *i* former customers, where $i \le n - 1$. It corresponds to the weighted sum of the profits over unidentified and identified customers, with weights δ^{n-i} and $(1 - \delta^{n-i})$ respectively; that is, for i < n,

$$\pi(i,n) = \frac{1-\beta}{r} \begin{bmatrix} \delta^{n-i} [\theta_i - \max\{\theta_{i+1}, p^u(i,n)\}] p^u(i,n) + \\ (1-\delta^{n-i}) \left(\frac{\theta_i^2}{2} - \frac{\theta_{i+1}^2}{2}\right). \end{bmatrix}$$
(1)

Straightforwardly, given θ_i and $\theta_{i+1} < \theta_i$, in any period n > i, the maximization of $\pi(i, n)$ with respect to $p^u(i, n)$ yields

$$p^{u*}(i,n) = \max\left\{\frac{\theta_i}{2}, \theta_{i+1}\right\}.$$
(2)

If $(1/2)\theta_i \leq \theta_{i+1} < \theta_i$, that is, if θ_{i+1} is not too far below θ_i , then the optimal price $p^{u*}(i, n)$ is equal to θ_{i+1} (the WTP of the lowest-valuation member of the vintage *i*), and all of them will purchase the good and enjoy a non-negative surplus *when they return* after their first-purchase. In contrast, if $(1/2)\theta_i > \theta_{i+1}$, then the optimal price is set at $(1/2)\theta_i$ and thus not all unidentified customers in vintage *i* will buy the good. Indeed, in those circumstances, customers with $\theta \in [\theta_{i+1}, (1/2)\theta_i]$ will not buy the good. Notice that in either case, vintage *i* marginal customers (those whose type is θ_{i+1}) will have zero surplus in all periods $n \geq i + 1$.

Accordingly, substituting (2) into Equation (1) yields the profit earned in period *n* from former customers of vintage *i*, where $i \le n - 1$,

$$\pi^{F}(i,n) = \frac{1-\beta}{r} \begin{bmatrix} \delta^{n-i} \left(\frac{\theta_{i}^{2}}{4}\right) + \\ (1-\delta^{n-i}) \left(\frac{\theta_{i}^{2}}{2} - \frac{\theta_{i+1}^{2}}{2}\right) \end{bmatrix} \text{ if } \frac{\theta_{i}}{2} \ge \theta_{i+1},$$

$$= \frac{1-\beta}{r} \begin{bmatrix} \delta^{n-i} (\theta_{i} - \theta_{i+1}) \theta_{i+1} + \\ (1-\delta^{j-i}) \left(\frac{\theta_{i}^{2}}{2} - \frac{\theta_{i+1}^{2}}{2}\right) \end{bmatrix} \text{ if } \frac{\theta_{i}}{2} \le \theta_{i+1},$$
(3)

where the superscript F is meant to clarify that we are concentrating on the profits that the monopolist makes on former customers (in period i).

The monopolist's profit in period *n* over all the former customers of all vintages, i = 0, 1, 2, ..., n - 1, is obtained by summing over *i*:

$$\Pi_{n}^{F} = \sum_{i=0}^{n-1} \pi^{F}(i, n).$$
(4)

Note that Π_n^F is a function of $\theta_{0,\theta_1,\theta_2,\dots,\theta_n}$, where $\theta_0 = 1$.

4.2 | Monopoly pricing for first-time customers

We now consider period *n* new customers. Let us denote by $p^{u}(n, n)$ the introductory price offered to new customers in period *n*. Let $U_{n}(\theta_{n+1})$ denote the intertemporal net utility of a period-*n* marginal new customer.²² By definition, she is indifferent between (i) becoming a first-time customer in period *n*, buying at the introductory price $p^{u}(n, n)$, and (ii) becoming a first-time customer in period n + 1, paying a lower introductory price $p^{u}(n + 1, n + 1)$ intended for period (n + 1) customers, but at the cost of foregoing a net utility $\theta_{n+1} - p^{u}(n, n)$ at each instant of time for the entire time interval Δ .

Under rational expectations, any period *n* first-time customer of type $\theta \in [\theta_{n+1}, \theta_n]$ anticipates correctly that in all later periods j > n, she will either (i) be offered an instantaneous price $p^{u*}(n, j) = \max\{\frac{\theta_n}{2}, \theta_{n+1}\}$ if her WTP remains unknown to the firm, or (ii) be charged the personalized price $p^I(\theta) = \theta$ if the monopolist has learnt her exact WTP. Consequently the period *n* marginal customer expects zero surplus in all later periods, implying that $U_n(\theta_{n+1})$ is equal to her net utility in period *n*, that is, $U_n(\theta_{n+1}) = \frac{1-\beta}{r}(\theta_{n+1} - p^u(n, n))$. It is convenient to express this equality as

$$p^{u}(n,n) = \theta_{n+1} - \frac{r}{1-\beta} U_{n}(\theta_{n+1}).$$
(5)

The monopolist's profit earned in period n from the period n first-time customers is

$$\prod_{n}^{N} = \frac{1-\beta}{r}(\theta_{n}-\theta_{n+1})p^{u}(n,n),$$

where the superscript *N* captures the fact that we are focusing on the profit made in period *n* over *N* ew customers. Using eq. (5) to substitute for $p^{\mu}(n, n)$, we obtain

$$\prod_{n=1}^{N} = \frac{1-\beta}{r} (\theta_n - \theta_{n+1}) \left(\theta_{n+1} - \frac{r}{1-\beta} U_n(\theta_{n+1}) \right).$$
(6)

If the firm wants to induce a new customer with $\theta \in [\theta_{n+1}, \theta_n]$ to buy in period *n* it must ensure that she would not be better off waiting until period *n* + 1. This participation constraint may be written as

$$U_n(\theta) \ge \beta U_{n+1}(\theta) \text{ for all } \theta \in [\theta_{n+1}, \theta_n].$$
(7)

However, since $U_n(\theta) \ge U_n(\theta_{n+1})$ for all $\theta \in [\theta_{n+1}, \theta_n]$, constraint (7) is redundant if the constraint $U_n(\theta_{n+1}) \ge \beta U_{n+1}(\theta_{n+1})$ is satisfied. This condition holds with equality because, by definition, a marginal customer is indifferent between being the first time customer in period *n* and being a first time customer in period n + 1.²³ It is thus equivalent to:

$$U_{n}(\theta_{n+1}) = \beta \left[\frac{(1-\beta)(\theta_{n+1}-\theta_{n+2})}{r(1-\delta\beta)} + U_{n+1}(\theta_{n+2}) \right] \text{if } \theta_{n+2} \ge \frac{\theta_{n+1}}{2},$$

$$U_{n}(\theta_{n+1}) = \beta \left[\frac{1-\beta}{r} \left(\theta_{n+1}-\theta_{n+2} + \frac{(\delta\beta)\theta_{n+1}}{2(1-\delta\beta)} \right) \right] \text{if } \theta_{n+2} \le \frac{\theta_{n+1}}{2}.$$
(8)
$$+U_{n+1}(\theta_{n+2})$$

The proof is in Appendix A.

5 THIRD-STAGE: THE MPE

We are dealing with a dynamic game between the monopolist and the consumers. We denote the state variable of our model by $\Theta(n)$, where $\Theta(0) = \theta_0 = 1$, and $\Theta(n) \equiv 1 - X(n)$, with $X(n) \in [0, 1]$ denoting the fraction of the total population that has purchased the good before period *n*. Because of the uniform distribution in [0, 1], $X(n) = 1 - \theta(n)$. It follows that $\Theta(n) = \theta(n)$. The real interval ($\Theta(n), 1$] represents the set of customers who have not purchased the good before period *n*.

Since the firm's optimal pricing for former customers has already been solved, the monopolist's dynamic optimization problem reduces to choosing how to expand its market coverage (or equivalently how to optimally price the good for first-time customers). The firm's Markovian strategy consists of a cut-off rule ψ , such that $\Theta(n + 1)$ is given by $\Theta(n + 1) = \psi(\Theta(n)) \le \Theta(n)$.

Concerning consumers, those who have not purchased the good before period *n* must decide on the optimal period to enter the market. We assume that consumers have a Markovian expectations rule $\Omega(.)$ that predicts the life-time rent of the marginal first-time customer in period *n*, that is, $U_n(\theta_{n+1})$. Under the rational expectations hypothesis, their predictions are always correct, that is,

$$\Omega(\Theta(n)) = U_n(\theta_{n+1}), \tag{9}$$

where, using (8), $U_n(\theta_{n+1})$ may be expressed as a function of the whole path of the state variable induced by the monopolist's strategy.

A Markovian strategy $\psi(.)$ chosen by the monopolist is a best reply to the consumer expectations function $\Omega(.)$ if it yields a sequence of cut-off values θ_{n+1} that maximizes profits, starting from any pair $(n, \Theta(n))$, and condition (9) is satisfied by such a sequence. Given the uniform distribution assumption and the general structure of the problem, we conjecture that in equilibrium, the monopolist's optimal cut-off rule is a linear,

$$\Theta(n+1) = \psi(\Theta(n)) = \gamma \Theta(n), \text{ with } 1 \ge \gamma \ge 0,$$
(10)

and consumers' expectations function is also linear:²⁴

$$\Omega(\Theta(n)) = \lambda \Theta(n). \tag{11}$$

In what follows, we will to determine the pair (γ^* , λ^*) such that the cut-off rule

$$\psi(\Theta(n)) = \gamma^* \Theta(n)$$

is the best reply to the expectations rule

$$\Omega(\Theta(n)) = \lambda^* \Theta(n)$$

and vice-versa.

The profit obtained in period n from old customers is given by Equation (4). Corresponding to Equation (6), the profit obtained in period n from first-time customers in that period is

$$\Pi_n^N(\Theta(n),\Theta(n+1),\lambda) = (\Theta(n) - \Theta(n+1)) \left(\frac{1-\beta}{r}\Theta(n+1) - \lambda\Theta(n)\right),\tag{12}$$

where we have substituted $\lambda \Theta(n)$ for $U_n(\theta_{n+1})$ because of the rational expectations requirement.

It follows that the Bellman equation for the monopolist is²⁵

$$V(\Theta(n)) = \max_{\Theta(n+1)} \left\{ \Pi_n^F + \Pi_n^N(\Theta(n), \Theta(n+1), \lambda) + \beta V(\Theta(n+1)) \right\}.$$
(13)

The results in Lemma 1 show that the consumers' best reply to the monopolist's cut-off exhibits two regimes, depending on the value of γ . In what follows, it is useful to refer to $1 \ge \gamma \ge 1/2$ as the case of slow market expansion or "market expansion by small steps" (as it implies that θ_{n+2} is not too far below θ_{n+1}). Analogously, $\gamma < 1/2$ may be called the case of "market expansion by large steps."

Lemma 1 (Consumers equilibrium expectations). Given any pair (β, δ) where $0 \le \beta \le 1$ and $0 \le \delta \le 1$, in any *MPE in which the monopolist's optimal cut-off rule and consumers' expectations rule are both linear in the state variable, it holds that the consumers' best reply to the monopolist's cut-off choice \gamma is given by*

$$r\lambda = \Lambda(\gamma, \beta, r) \text{ if } 1 \ge \gamma \ge \frac{1}{2},$$

$$r\lambda = \Lambda(\gamma, \beta, r) - \frac{\beta^2 (1 - \beta)\gamma \delta (1 - 2\gamma)}{2(1 - \beta\gamma)(1 - \beta\delta)} \text{ if } \gamma \le \frac{1}{2},$$
(14)

with $\Lambda(\gamma, \beta, \delta)$ given by

$$\Lambda(\gamma,\beta,\delta) \equiv \frac{\beta\gamma(1-\beta)(1-\gamma)}{(1-\beta\gamma)(1-\beta\delta)}.$$
(15)

 \square

Proof. See Appendix A.

Simple computations show that, for a given market expansion strategy of the monopolist, the expected lifetime utility of the marginal first time customers is increasing in δ ; meaning that it is the greater the more slowly the firm learns the preferences of its past customers. The intuition for this result is rather clear. The higher is δ , the greater is the expected surplus that a consumer may obtain by delaying her purchase to the next period and hence the higher the informational rent she must be offered to buy in the present period. This is because the probability of retaining a positive surplus in all future periods by delaying one's first purchase is increasing in δ . Indeed the more slowly the firm learns about its past customers' preferences, the smaller is the fraction of customers against which it will be able to

exercise first-order price discrimination and the greater the fraction who will be offered a vintage-dependent price (third-order price discrimination).

Taking the expectations parameter λ as given, the monopolist chooses the optimal cut off rule.

Lemma 2 (Monopolist's Optimal Market Expansion). Given any pair (β, δ) where $0 \le \beta \le 1$ and $0 \le \delta \le 1$, in any MPE in which the monopolist's optimal cut-off rule and consumers' expectations rule are both linear in the state variable, it holds that the monopolist's cut-off choice γ is a best reply to the consumers' expectation rule iff the following conditions are satisfied:

$$h(\gamma, \beta, \delta) = 0 \text{ if } 1 \ge \gamma \ge \frac{1}{2},$$

$$k(\gamma, \beta, \delta) = 0 \text{ if } \gamma \le \frac{1}{2},$$
(16)

where $h(\gamma, \beta, \delta)$ and $k(\gamma, \beta, \delta)$ are defined by Equations (A16) and (A17).

Proof. See Appendix A.

Lemma 2 implies that, for given consumers' expectations (i.e., a given value of λ), at least in the case of expansion by small steps ($1 \ge \gamma \ge 1/2$), market expansion is going to be quicker when the monopolist learns more quickly the preferences of its past customers.²⁶ This is simply that because it is generally more profitable to expand quickly the market when the proportion of customers who become identified is greater at any point in the future. The exception arises only when the market expands fast (the monopolist expands it in large steps) and γ is too small. In that case, there is possibly a countervailing effect since the proportion of unidentified customers of each vintage who don't return buying may be large.

The following result follows immediately:

Proposition 1. Given any pair (β, δ) where $0 \le \beta \le 1$ and $0 \le \delta \le 1$, there exists an MPE in which the monopolist's optimal cutoff rule and consumers' expectations rule are both linear in the state variable, where the equilibrium parameters (γ^e, λ^e) are such that:

- (i) The monopolist's equilibrium cut-off value $\gamma^e(\beta, \delta)$ is either as a root of Equation (A18), with $1 \ge \gamma \ge 1/2$ or a root of Equation (A19), with $1/2 \ge \gamma \ge 0$,
- (ii) $\lambda^{e}(\beta, \delta)$ is given by Equation (A6).

Proposition 1 allows us to investigate how the nature of the market expansion depends on the three parameters μ , r and Δ , because the equilibrium values of γ and λ can be equivalently expressed as $\gamma^e(r, \mu, \Delta)$ and $\lambda^e(r, \mu, \Delta)$, due to the definitions $\beta \equiv e^{-r\Delta}$ and $\delta \equiv e^{-\mu\Delta}$. When we focus on the effects of a variation in the degree of precision (or coarseness) of information, it will be convenient to study the effects of variations in the value of μ , for given values of r and Δ . This is equivalent to study the effect of variations of δ for a given value of β . On the contrary, when we try to ascertain the effects of a variation in the length Δ of the period of contractual commitment, we shall consider fixed values of r and μ (i.e., a given degree of precision of the monopolist's information).

It is also important to note that, as shown in Lemma 1, Lemma 2, and Proposition 1, the economic properties of the monopolist's optimal expansion strategy and consumers' expectations are indeed different, depending on whether, in equilibrium, γ^e is above or below $\frac{1}{2}$. Thus it is useful to try to characterize the circumstances leading to the razor edge case in which the equilibrium γ^e is exactly equal to $\frac{1}{2}$, which defines the threshold determining the monopolist's optimal decision to go for a slow or a fast market expansion regime. Remarks 1 and 2 provide additional insights on this matter.

Remark 1.

(i) For any given β in (0, 1), the monopolist will choose to expand the market in large steps (i.e., $\gamma < 1/2$) when the degree of precision of information on customers is high enough (i.e., when δ is small enough),

(ii) There exists a value $\overline{\delta} = \Phi(\beta) < 1$ such that, for any δ in $(0, \overline{\delta})$, the monopolist will choose to expand the market in large steps (i.e., $\gamma < 1/2$) if β is big enough (i.e., if *r* or Δ is small enough).

Proof. See Appendix A.

Remark 2. For any pair $(r, \mu) > (0, 0)$, in the limiting case where Δ becomes arbitrarily small, the whole market is covered instantaneously (Figure 3 confirms this finding).

Proof. See Appendix A.

In Figure 1, we have plotted the equilibrium value γ^e of the cutoff parameter as a function of the degree δ of coarseness of information for $\beta = 0.8$.²⁷ The green part of the curve corresponds to values of $\delta \in [0, 0.12848]$ which ensures that $\gamma^e(0.8, \delta) \leq 1/2$ (market expansion by large steps).²⁸ It turns out that γ^e decreases when δ decreases, that is, the equilibrium market expansion proceeds in bigger steps if the precision of information is higher (as mentioned in (i) of Remark 1). This is an intuitively plausible result for two different reasons. On one hand, as noticed after Lemma 2, the faster the fraction of unidentified former customers shrinks, given any consumers' expectations rule, the more the monopolist is interested in expanding quickly its market (given the prospects of additional profits resulting from first-degree price discrimination over a considerable set of customers). On the other hand, as noticed after Lemma 1, the greater the probability that the firm learns its customers' exact WTP, the smaller the rents they require not to delay their first purchases and, accordingly, the greater the incentive for the firm to expand quickly its market.

Figure 2 illustrates the results in part (ii) of Remark 1. It depicts the relationship between γ^e and β for two different values of the coarseness parameter, $\delta = 0.8$ (the upper curve) and $\delta = 0.1$ (the lower curve). As shown in Figure 2, if the information is coarse enough (large values of δ), then γ^e is greater than 1/2 (market expansion by small steps) for all values of β . For small values of δ , γ^e becomes smaller than 1/2 if β is sufficiently large (the green part of the lower curve).

Let us now consider the relationship between the speed of market expansion and the length of the period of contractual commitment. Recall that the monopolist adopts a linear market expansion rule: in any period *n*, the fraction of the customer base that is not served is $(\gamma^e(\beta, \delta))^{n+1}$. The measure of customers that have been served by the end of period *n* is $(1 - (\gamma^e)^{n+1})$. For any time $t = n\Delta$, we have $(\gamma^e)^{n+1} = (\gamma(\Delta))^{\frac{t}{\Delta}+1}$. The following result, stated as Claim 1, has a Coasian flavor, as it shows that, whatever the value of $\mu > 0$, in the limit when $\Delta \to 0$, the market is covered instantaneously.

Claim 1. When the length of time Δ between two different proposals to two consecutive sets of new customers becomes infinitesimal, the market is covered instantaneously. For any given time t > 0, as the length of the period of commitment Δ tends to 0 the fraction of the customer base that has been served up to that time tends to $1 - e^{\gamma'(0)t}$ where $\gamma'(0) \rightarrow -\infty$

$$\lim_{\Delta \to 0} (\gamma(\Delta))^{\frac{t}{\Delta}+1} = e^{\gamma'(0)t} = 0,$$

so that the market is covered by the monopolist in a twinkling of an eye at the beginning of the game.



FIGURE 1 $\gamma^{*}(0.8, \delta)$ [Color figure can be viewed at wileyonlinelibrary.com]

FIGURE 2 γ^* as a function of β [Color figure can be viewed at wileyonlinelibrary.com]



FIGURE 3 γ^* as a function of Δ [Color figure can be viewed at wileyonlinelibrary.com]

The result in Claim 1 is driven by the firm's interest in expanding the market as fast as possible when Δ tends to zero, as in standard Coasian dynamics. In that limiting case, we have $\delta \rightarrow 1$, regardless of the value of μ . Thus, $\delta > \Phi(\beta)$ for very small Δ , implying that $\gamma^e > 1/2$. Even though the market expands by small steps in each period, since the periods are becoming very short when Δ tends to zero, it holds that, between any two given points of time t_1 and $t_2 > t_1$, the number of customers served by the firms is larger, the smaller is Δ . Even if, in each period the firm may not get information on a large fraction of customers, as periods last for infinitesimally short amount of time, the firm is able to get much information on customers' WTP in a short amount of time. Thus, the firm can quickly engage in personalized prices that allow it to extract, for each individual, all the consumer surplus generated by each transaction. This means that, in the MPE, the firm expands very fast when Δ is very small, leading to instantaneous market coverage in the limit case where $\Delta \rightarrow 0$.

In Figure 3 we picture γ^e as a function of the length Δ of the period of contractual commitment, where we have set r = 1 and $\mu = 0.5$. Notice that in Figure 3, for all $\Delta > 1$, we have $\gamma^e > 1/2$ (expansion by small steps). Consistent with Claim 1, Figure 3 shows that the slope of this curve tends to minus infinity as Δ tends to zero.

6 ECONOMIC EFFECTS OF CUSTOMERS' PROFILING

6.1 Profits and information precision: A U-shaped relationship

To identify the profit effects of using customers' data to engage in sophisticated price discrimination practices (including first-degree price discrimination over a subset of customers, whose WTP is uncovered by the monopolist),

we now consider the relationship between the monopolist's life-time profit, denoted by Π^* , and the parameters β and δ , where

$$\Pi^* \equiv \sum_{n=0}^{\infty} \beta^n \Big[\Pi_n^F + \Pi_n^N \Big]$$

and Π_n^F and Π_n^N are defined by (6) and (4), and where $\theta_{n+1} = \gamma^e(\beta, \delta)\theta_n$. Letting $\gamma = \gamma^e(\beta, \delta)$ we obtain the monopolist's life-time profit,

$$\Pi^{*}(\beta,\delta) = \frac{(1-\gamma)[2\gamma+\beta^{2}\gamma(1+\gamma(\delta-1)+\delta)-\beta(\gamma(3+\delta)+\delta-1)]}{2r(1-\beta\gamma)(1-\beta\gamma^{2})(1-\beta\delta)},$$
(17)

for all (β, δ) such that $\gamma^e(\beta, \delta) \ge 1/2$, and

$$\Pi^{*}(\beta,\delta) = \frac{\beta\delta\{-1-\beta+[\beta(7-\beta)-4]\gamma+2[3+(\beta-3)\beta]\gamma^{2}-2\beta\gamma^{3}\}}{4r(1-\beta\gamma)(1-\beta\gamma^{2})(1-\beta\delta)}$$
(18)

for all (β, δ) such that $\gamma^e(\beta, \delta) \le 1/2$. (Clearly, the two expressions are equal at $\gamma = 1/2$, so that the monopolist's lifetime profit is a continuos function of γ).

The relationship between the monopolist's life-time profit and the degree of coarseness of information is drawn in Figure 4a for $\beta = 0.8$ (and r = 1).²⁹ Cases $\beta = 0.5$, r = 1 and $\beta = 0.2$, r = 1, are pictured in Figures 4b,c. Please note that Figure 4a–c shows the same qualitative relationship between profit and δ (interpreted here as the degree of coarseness of the firm's customer information). When δ approaches 1, (i.e., the information on previous customers is very coarse, revealing only the date that a consumer makes her first purchase and the monopolist can never discover former consumers' exact WTP), the monopolist is able to engage in third-degree price discrimination among different vintages of customers but profit is at its minimum. If the monopolist has strictly no information on its former customers (not even when or whether they have purchased the good), the only possible equilibrium is the repetition of the static monopoly equilibrium, leading to equilibrium profit equal to 0.25/r. This obviously implies that the monopolist's life-time profit is non-monotonic with respect to the degree of information ($\delta = 1$ or, equivalently, $\mu = 0$) and then continuously increases when the degree of precision of the collected information increases, taking its maximum value when $\delta = 0$ ($\mu = +\infty$).

In the scenario in which it cannot obtain any information at all about former customers, the monopolist's optimal pricing is to charge a time-invariant price, p = 1/2 and only 1/2 of the market is served, leading to a life-time profit of 0.25/r. When the technology enables the collecting and processing of information on former customers, the monopolist, being unable to commit on future prices, has the incentive to practice intertemporal price discrimination, offering successively lower and lower introductory prices to previously unserved customers, hoping to gain information about them once they have bought the good. If this information is coarse, that is, δ is close to 1, meaning that after each period, the fraction of former customers whose exact WTP cannot be identified is near 1 (as in the baseline model of Laussel et al., 2020a), the monopolist's life-time profit becomes strictly lower than 0.25/r, because consumers, expecting lower and lower future introductory prices, prefer to postpone their purchases unless they are offered significant information on customers' preferences at all (in which the static equilibrium would be repeated indefinitely). Life-time profit is greater than 0.25/r only if the consumers' profiling technology is sufficiently advanced, allowing δ to fall below a critical threshold level δ_c . (e.g., in Figure 4a, $\delta_c \simeq 0.84$). Profit attains its global maximum at the polar case $\delta = 0$, in which the monopolist gets information on customers' WTP (and price discriminates accordingly), immediately after the first purchase.

These qualitative results appear to be robust with respect to the value of β . In Figure 4b,c, the same qualitative relationship is obtained for some other values of β . There is always a threshold value δ_c (which depends on β), such that if $\delta \in (\delta_c, 1]$ then life-time profit is below 0.25/r. This is true for all values of $\beta \in (0, 1]$. As is obvious from Figure 4b,c, the curve is more and more horizontal for smaller and smaller values of β , converging to the horizontal dashed line as $\beta \to 0$. When $\beta = 0$ (which means $\Delta = \infty$) the "first commitment period" lasts for ever, and hence, not surprisingly, the equilibrium life-time profit is the same as that obtained in the case of infinite repetition of the static equilibrium.



FIGURE 4 (a) $\Pi^*(0.8, \delta)$, (b) $\Pi^*(0.5, \delta)$, and (c) $\Pi^*(0.2, \delta)$ [Color figure can be viewed at wileyonlinelibrary.com]

Let us now consider the special case where $\Delta = 0$ (so that $\beta = \delta = 1$). For this purpose, let us express life-time profits as a function of Δ for a given pair (r, μ) , $\Pi^*(\Delta; r, \mu) = \Pi^*(e^{-r\Delta}, e^{-\mu\Delta})$ where Δ is the length of the period of contractual commitment. Claim 2 shows the limiting behavior of profit as $\Delta \to 0$.

Claim 2. When the length of time Δ between two different proposals to two consecutive sets of new customers becomes infinitesimal, $\Pi^*(\Delta; r, \mu) \rightarrow \frac{1}{2r} \frac{\mu}{r+\mu}$. Accordingly $\Pi^*(0; r, \mu)$ is increasing in the degree μ of precision of information, varying from $\Pi^*(0; r, \mu) = 0$ when $\mu = 0$ (i.e., when the only information which the monopolist gets on his former customers is their date of first purchase) to $\Pi^*(0; r, \mu) = \frac{1}{2r}$ when $\mu \rightarrow \infty$ (i.e., when the monopolist learns instantaneously the exact WTP of his consumers). Notice that $\frac{1}{2r}$ is the monopolist's life-time profit when he can fully discriminate among all consumers (first-degree price discrimination).

Using Claim 2, we can prove the following Corollary, which compares the monopolist's life-time profit in the limiting case $\Delta \to 0$ with that in the limiting case $\Delta \to \infty$, for given *r* and μ . It turns out that the former is greater than the latter if $\mu < r$ and the opposite holds if $\mu > r$.

Corollary 1. The monopolist's life-time profit when $\Delta \to \infty$ is greater (smaller) than when $\Delta \to 0$ iff r is greater (smaller) than μ .

SIGN
$$[\Pi^*(+\infty; r, \mu) - \Pi^*(0; r, \mu)] = SIGN [r - \mu].$$

Proof. It suffices to notice that: (i) from Claim 2, we have $\Pi^*(0; r, \mu) = \frac{1}{2r} \frac{\mu}{r+\mu} \left(\ge \frac{1}{4r} \Leftrightarrow \mu \ge r \right)$, and that (ii) from (17) or (18) and $\gamma^e(0, 0) = 1/2$, we have $\Pi^*(+\infty; r, \mu) = \frac{1}{4r}$ for all $\mu \ge 0$.

Corollary 1 states a striking result: the dynamics of this model is globally Coasian if $r > \mu$ whereas it is globally non-Coasian if $r < \mu$. In the first case, the firm's life-time profit when the length of the commitment period tends to 0 is smaller than its life-time profit when this length is infinite; that is, if information is coarse (small μ), the monopolist would be better off by making an initial fixed price offer that lasts for ever (selling only to the top half of the customer base) than offering a sequence of prices (each lasting just an instant) which only helps discover very slowly customers' exact WTPs. This is because the future gains in profits from first-degree price discrimination grow only slowly and are heavily discounted (high *r*).

In the second case, that is, $\mu > r$, the reverse holds: offering a sequence of prices (each lasting just an instant) helps discover rather quickly (thanks to a large μ) customers WTP, and the future gains are not heavily discounted (thanks to a low *r*), leading to a larger life-time profit than that obtained by making an initial fixed price offer that lasts for ever. Notice however that Corollary 1 compares the values of the profit for the two extreme values of Δ ; and in general $\Pi^*(\Delta; r, \mu)$ is not a monotonic function of Δ .³⁰

This shows that in this model there are two effects working in opposite directions. The first one (which is the stronger whenever $r > \mu$) is the classical Coasian effect: when introductory prices are fixed for a shorter period (Δ is smaller), consumers expect lower future introductory prices to occur more quickly, and at the same time they bear a lower waiting cost so that they prefer to postpone their purchases, implying that the prices decrease even more quickly. The second effect (which is the stronger whenever $r < \mu$) is a "front-loading effect"³¹: when Δ is smaller, the monopolist is able to extract sooner the surplus of traceable former customers whose exact WTP becomes known to the monopolist.

6.2 Consumers surplus and information precision: A hump-shaped relationship

In this subsection, we investigate how the degree of accuracy on consumers' profiling impacts social welfare and consumer surplus.

Let w(n) denote the social welfare at the MPE in period *n*. As usual, we define social welfare as the sum of consumer surplus and profit. Consumer surplus is gross utility minus the payment made to the firm. Profit is the sum of consumers' payments to the firm (because we assume that the cost of production is zero). Therefore welfare in any period is simply the gross utility of consumers who are served in that period. Whenever $\gamma^e(\beta, \delta) \ge 1/2$, all customers whose type $\theta \ge \theta_{n+1}$ are served in period *n*. Thus the social welfare that accrues in period *n* is

$$w(n) = \frac{1-\beta}{2r} \left(1-\theta_{n+1}^2\right) = \frac{1-\beta}{2r} (1-(\gamma^{n+1})^2).$$
(19)

When $\gamma^e(\beta, \delta) \leq 1/2$, things are slightly more complicated, as not all new customers whose type belongs to the interval $(\theta_{n+1}, \theta_n]$ are served in period *n*. Consider indeed vintage *i* customers, i < n. A fraction $(1 - \delta^{n-i})$ of them are served whatever their type, because their individual WTP have been exactly identified and the monopolist is able to make a personalized (limit) offer to each of them. For the remaining fraction δ^{n-i} , only the types between $\frac{\theta_i}{2}$ and θ_i are served as the monopolist prefers to leave some of the consumers in this vintage unserved (to extract more surplus from returning shoppers). Thus the social welfare that accrues in period *n* is, in that case:

$$w(n) = \frac{1-\beta}{2r} \left[\left(\theta_n^2 - \theta_{n+1}^2\right) + \sum_{i=0}^{n-1} \left((1-\delta^{n-i}) \left(\theta_i^2 - \theta_{i+1}^2\right) + \delta^{n-i} \frac{3}{4} \theta_i^2 \right) \right].$$
(20)

The overall aggregate social welfare is the sum of the present values of w(n) over all n, that is;

$$W(\beta, \delta) = \sum_{n=0}^{\infty} \beta^n w(n) = \frac{1}{2r} \frac{1 - \gamma^2}{1 - \beta \gamma^2},$$
(21)

when $\gamma = \gamma^e(\beta, \delta) \ge 1/2$ and

$$W(\beta,\delta) = \sum_{n=0}^{\infty} \beta^n w(n) = \frac{4 + (\beta - 5)\beta\delta - 4\gamma^2 (1 + (\beta - 2)\beta\delta)}{8r(1 - \beta\gamma^2)(1 - \beta\gamma)},$$
(22)

when $\gamma = \gamma^e(\beta, \delta) \le 1/2$.

Notice that the two values are equal whenever $\gamma^e(\beta, \delta) = 1/2$. Very intuitively, social welfare is greater, the finer is the monopolist's information about consumers' preferences. This is quite obvious in the case $\gamma^e(\beta, \delta) \ge 1/2$. Indeed, from (21), social welfare is the greater the quicker is the market expansion (i.e.; the smaller is $\gamma^e(\beta, \delta)$) and we have already seen, as illustrated in Figure 1, that market expansion proceeds by larger steps (smaller γ) the more precise is the firm's information (i.e., the smaller is δ). This holds true as well in the case $\gamma^e(\beta, \delta) < 1/2$. Figure 5 illustrates this point, showing social welfare as a function of δ (for $\beta = 0.8$ and r = 1).

Notice that if the monopolist does not obtain any information on previous customers and repeats forever the static monopoly equilibrium, then social welfare equals $\frac{3}{8r}$ which is greater than $W(\beta, 1)$ but smaller than $W(\beta, 0)$. Thus the relationship between social welfare and the degree of precision of information is also non-monotone. Notice that in the limit case when $\Delta \rightarrow 0$, social welfare $\rightarrow \frac{1}{2r}$ whenever the firm has some information about its former customers, whatever the degree μ of precision of this information. It is accordingly always better in that case, from the social welfare point of view, that the monopolist be informed, even in the coarsest possible way.

What about the aggregate consumers' surplus $CS(\beta, \delta)$? Since it equals the difference between social welfare and profits, we easily obtain it as

$$CS(\beta,\delta) = \frac{(1-\beta)(1-\gamma)^2(1+\beta\gamma)}{2r(1-\beta\gamma)(1-\beta\gamma^2)(1-\beta\delta)},$$
(23)

when $\gamma = \gamma^e(\beta, \delta) \ge 1/2$ and

$$CS(\beta, \delta) = \frac{(1 - \beta)(4(1 - \gamma)^2(1 + \beta\gamma))}{8r(1 - \beta\gamma)(1 - \beta\gamma^2)(1 - \beta\delta)},$$
(24)

when $\gamma = \gamma^*(\beta, \delta) \ge 1/2$.

The relationship between *CS* and δ is pictured in Figure 6 for $\beta = 0.8$ and r = 1.



FIGURE 5 $W(0.8, \delta)$ [Color figure can be viewed at wileyonlinelibrary.com]



FIGURE 6 $CS(0.8, \delta)$ [Color figure can be viewed at wileyonlinelibrary.com]

Without any information about former customers, the infinite repetition of the static monopoly equilibrium leads to a consumer surplus equal to $\frac{1}{8r}$. When the firm acquires the coarsest possible information (corresponding to $\delta = 1$), the consumers surplus jumps to a greater level before decreasing when δ decreases. The relationship between *CS* and the degree of precision of information turns out to be in the opposite of the relationship between monopoly profit and information precision: it is hump-shaped, with a maximum value attained when the firm's information is the coarsest one ($\delta = 1 \Leftrightarrow \mu = 0$) This was to be expected since the firm uses information on consumer's profiling to set an individual price that coincides with the consumers' WTP, fully extracting the associated consumer surplus. When firms' ability to segment consumers according to their tastes remains rather coarse, consumers actually benefit from price discrimination (at least those whose preferences remain uncovered) as they are able to extract information rents for successive periods (until their preferences get disclosed).

In the limit case when $\Delta \to 0$, CS tends toward $\frac{1}{2r} \frac{r}{r+\mu}$. In that case, it is interesting to notice that, when the firm is able to obtain some information about its old customers, the social welfare is constant with μ , with profit and consumer surplus effects offsetting each other: the share of profits is increasing in μ from 0 to 1, and the share of consumers surplus is decreasing in μ from 1 to 0.

7 SECOND-STAGE: OPTIMAL INVESTMENT IN CUSTOMERS PROFILING

In the previous sections, we analyzed the MPE of the game for a given value of the profiling accuracy μ , showing that the greater is the profiling accuracy the greater are the equilibrium profits, keeping in mind that the profits in the coarsest information case ($\mu = 0$) are smaller than in the no information case. However, the accuracy of the information obtained by the firm is neither exogenous nor costless. It clearly depends on the amount the firm is willing to invest to collect and exploit consumers data to acquire the useful hardware and software, to hire good programmers, chief data officers and so on. Let us then look at the second-stage of our game, where we endogenize the firm's decision about the degree of information precision on customers tastes. We simply assume that the firm's investment in big data is an increasing function $F(\mu)$ of the information accuracy which it wants to reach and that this investment must take place, once and for all, at the beginning the first period of the game (this assumption intends to capture that customers' profiling activities often require up-front investment in hardware and software whose costs are independent of the subsequent level of firms' engagement in customers profiling). We suppose in addition that $F(0) \ge 0$, $F'(\mu) \ge 0$ and F''(μ) ≥ 0 , meaning that (i) collecting even the coarsest information may be costly, (ii) the amount invested is increasing in the accuracy of information and (iii) it is increasing at an increasing rate.

Accordingly the monopolist's net profits, when it chooses to invest in information collection, is the following function of μ :

$$\Pi^*(\Delta \mu, r) - F(\mu),$$

where $\Pi^*(\Delta \mu, r)$ follows from (17) and (18), with $\beta = e^{r\Delta}$, $\delta = e^{\mu\Delta}$.

When choosing to engage in customers profiling, the monopolist selects the value of μ to maximize its net profit, that is,

$$\mu^* = \arg \max_{\mu} [\Pi^*(\Delta; \mu, r) - F(\mu)],$$
(25)

which is equivalently expressed as

$$\delta^* = \arg \max_{\delta \in [0,1]} \left[\Pi^*(\beta, \delta) - F\left(-\frac{1}{\Delta} \ln \delta\right) \right].$$

As is well known, according to the Maximum Theorem, a continuous function always has a maximum over a closed interval.

Then, the firm chooses to become informed iff

$$\Pi^{*}(\Delta; \mu^{*}, r) - F(\mu^{*}) > \frac{1}{4r},$$
(26)

that is, iff this yields profits which are greater that the profits which it would earn by choosing not to collect any information at all on its customers.

Given the complexity of the profit function, it is impossible to derive in the general case analytical results on the optimal value of μ and on the precise conditions which ensure that the monopolist chooses to invest in information collection. Nevertheless it is straightforward to see that when the firm does invest in customer profiling, it must always target an information accuracy above a lower bound value μ which is larger than 0 whenever Δ is finite.

Let us indeed formally define μ as the value of μ which solves $\Pi^*(\Delta; \mu, r) = \frac{1}{4r}$.

Claim 3. The optimal profiling accuracy $\mu^* \ge \mu$.

Proof. It is enough to remember that
$$\Pi^*(\Delta; \mu, r)$$
 is increasing in μ and that $F'(\mu) \ge 0$.

The intuition behind Claim 3 is simply that, even if collecting information is costless, the monopolist should optimally abstain from investments in information collection if such investment does not permit it to reach at least a minimum information accuracy level μ , which assures that the monopolist gets information on the true WTP of a minimal critical mass of customers, so that the additional future profits made on those customers compensate not only the investments in tracking and profiling technologies (which would depend on the up-front investments on hardware, software and human resources) but also the profit losses resulting from the monopolist's inter-temporal price discrimination incentives.

In what follows, for tractability, let us assume that the investment cost function takes the form $F(\mu) = f + k\mu$, where k > 0 and $f \ge 0.^{32}$ We call k the marginal investment cost of informational accuracy. If f > 0, then to obtain even the coarsest level of information, $\mu = 0$, which allows only purchase history information, the firm must incur a strictly positive fixed investment cost f (e.g., when $\mu = 0$, this parameter captures the costs to obtain and maintain a data set where the firm registers each customer and the time of the first purchase). On the other hand, if the firm chooses not to invest at all, then the cost f is avoided, and the firm will have no information whatsoever about individual customers (as mentioned earlier, in this case, we have a repetition of the static solution as all customers are perceived by the firm as first-time shoppers, regardless of whether they had actually bought the product in the past or not). The optimal information accuracy, conditional on the monopolist investing in big data, is then $\mu^*(k, \Delta, r)$, which is decreasing in k. Let us denote by $PR(\Delta; \mu^*(k, \Delta, r), r, k, f)$ the monopolist's net profits conditional on investing in big data. From the Envelope Theorem, we have $dPR/dk = -\mu$, thus we obtain

$$PR(\Delta; \mu^*(k, \Delta, r), r, k, f) = PR(\Delta; \mu^*(0, \Delta, r), r, 0, f) - \int_0^k \mu^*(\kappa, \Delta, r) d\kappa$$

Notice that if k = 0, then, conditional on investing, the optimal level of informational accuracy is $\mu^* = \mu^*(0, \Delta, r) = +\infty$, which implies $\delta = 0$. Thus $PR(\Delta; \mu^*(0, \Delta, r), r, 0, f)$ equal $\Pi^*(\beta, 0) - f$, where $\Pi^*(\beta, 0)$ are the firm's gross profits under full information acquisition. It follows that

$$PR(\Delta; \mu^*(k, \Delta, r), r, k, f) = \Pi^*(\beta, 0) - f - \int_0^k \mu^*(\kappa, \Delta, r) d\kappa.$$

Recall that $\Pi^*(\beta, 0) > 1/4r$, provided that $\beta > 0$. Clearly, if $f > \Pi^*(\beta, 0) - 1/4r$, then the right-hand side of the above equation is negative, and the firm would be better off by not investing in big data at all. Therefore, for investment to take place, it is necessary that $f \le \Pi^*(\beta, 0) - 1/4r$, and we will focus on this case. Then, for each f in the interval $[0, \Pi^*(\beta, 0) - 1/4r]$, there is a corresponding marginal investment cost parameter value $\overline{k}(f) \ge 0$ such that $PR(\Delta; \mu^*(\overline{k}, \Delta, r), r, \overline{k}, f) = 0$, implying that the monopolist would find it profitable to invest in big data iff $k \le \overline{k}(f)$.³³

Since μ^* is a decreasing function of k, there is a critical value $\mu^{Crit}(f) \equiv \mu^*(\overline{k}(f), \Delta, r)$ such that whenever the monopolist finds it strictly profitable to invest in data collection (i.e., whenever $k < \overline{k}(f)$), its optimal accuracy level, $\mu^*(k, \Delta, r)$, is greater than $\mu^{Crit}(f)$. Thus we have proved the following claim:

Claim 4. Profitable investments in data collection requires to invest an amount of money greater than some critical value $F(\mu^{Crit}(f)) > 0$.

The following simple example, when $\Delta \rightarrow 0$, may be instructive.

Example 1. Let $\Delta = 0$, so that $\Pi^*(0; \mu, r) = \frac{1}{2r} \frac{\mu}{r+\mu}$. Straightforwardly $\underline{\mu} = r$ and $\mu^* = \max\left\{\frac{1}{\sqrt{2k}} - r, 0\right\}$. Clearly, the firm cannot benefit from collecting information if $r \ge \frac{1}{\sqrt{2k}} \Leftrightarrow k \ge \frac{1}{2r^2}$.³⁴ Consider then the case where $k < \frac{1}{2r^2}$. Then $\Pi^*(0; \mu^*, r) - F(\mu^*) = \frac{1}{2r} + rk - \sqrt{2k} - f$. Condition (26) is then equivalent to

$$f < \frac{1}{4r} + rk - \sqrt{2k},\tag{27}$$

or, equivalently,

$$k \le \bar{k}(f) = \frac{3 + 4rf - 2\sqrt{2 + 8rf}}{4r^2}.$$
(28)

Since *k* cannot be negative, Equation (28) means that the maximum possible value of *f* that is consistent with profitable investment is $\frac{1}{4r}$. It is easy to see that $\overline{k}(f) < \frac{1}{2r^2}$ so that (28) is necessary and sufficient to ensure that the monopolist invests in profiling. From Equations (27) or, equivalently (28), the monopolist is all the more likely to benefit from collecting actively consumers' information as the fixed and variable costs of information collection, *f* and *k*, are small and the discount rate is small itself, what is hardly surprising.

Notice that, to profitably invest in data collection, the monopolist must aim at reaching an accuracy level greater than some critical value. Let us find what this minimum value is. Remembering now that $\mu^* = \frac{1}{\sqrt{2k}} - r$ and using (28) we find that when a profitable investment in big data takes place μ^* has to be greater than

$$\mu^{Crit} = r \left(\frac{\sqrt{2}}{\sqrt{3 + 4rf - 2\sqrt{2 + 8rf}}} - 1 \right).$$
(29)

This is an increasing function of f which equals $r\left(\sqrt{\frac{2}{3-2\sqrt{2}}}-1\right)$ when f=0 and tends toward ∞ when $f \to \frac{1}{4r}$. It is in any case strictly greater than $\underline{\mu} = r$.

8 FIRST-STAGE: STRATEGIC CONSUMERS

Consumers' equilibrium surplus (CS) is impacted by the information which the monopolist is able to collect on its previous customers. We have seen that CS equals 1/8r when the firm has no information whatsoever about its previous customers, and that CS jumps to its maximum value when the firm obtains the coarsest possible information and then decreases when the information becomes more and more precise. Obviously the consumers have an incentive to influence the degree and the precision of the information which the monopolist can collect.

We now look at the first stage of the game, where consumers can take collective actions to maximize expected CS, through lobbying and/or political action which influences legislation on consumers data protection, thereby restraining firms' ability to collect more and more information on customers' preferences and WTP (e.g., through tighter RGPD legislation).³⁵ This action, which we assume for the sake of simplicity to be costless, takes place in the initial stage of the game, before the monopolist chooses the amount of money to invest in customers profiling. More precisely, we consider that consumers' action can raise the monopolist's cost of reaching any given level of information accuracy. Considering the simple case of the previous section where $F(\mu) = f + k\mu$, let us assume that in the absence of consumers' collective action, f and k take the value f_A and k_A . (The subscript A means "in the absence of protective legislation"). Consumers' action may result in $f \ge f_A$ and $k \ge k_A$.

The problem would be trivial if it is not profitable to invest in data profiling even in the absence of consumers data protection, that is, if $k_A > \overline{k}(f)$. So consider the nontrivial case where investing in data profiling is optimal absent any legislation. What can a protective legislation achieve? What equilibria can be reached by appropriately choosing the values of f and k above their natural values f_A and k_A ? Given the results in the previous section, possible equilibria include the no information one (yielding CS = 1/8r), which may be implemented by enacting a very restrictive data protection legislation (high enough values of f and k), and equilibria where the monopolist invests in profiling. Among the latter, for a given value of f, the consumer surplus is increasing in k (since a higher k implies a smaller $\mu^*(k)$). The highest value of k for which the firm finds it profitable to invest in profiling is $\overline{k}(f)$. This is accordingly the best "information equilibrium" from the consumers' point of view. The expected consumer surplus at this equilibrium is $CS((\Delta; \mu^{Crit}(f), r)$.

Claim 5. Suppose that $k_A \leq \overline{k}(f)$, so that investing in data collection is profitable in the absence of a protective legislation. Then:

- (i) When $1/8r \ge CS((\Delta; \mu^{Crit}(f), r))$, the best consumers' strategy is to favor a data protection legislation which dissuades any data collection by the monopolist;
- (ii) When $1/8r < CS((\Delta; \mu^{Crit}(f), r))$, the best consumers' strategy is to favor a value of $k = \overline{k}(f) \ge k_A$ such that the monopolist invests in profiling but the information accuracy remains low, with $\mu^{Crit}(f) < \mu^*(k_A)$; in that case the protective legislation does not prevent data collection but reduces the precision of the information which is collected.

Examples 1 (Continued). Assume for simplicity that the legislation can only impact k but not f. As already shown, when $\Delta \to 0$, $CS \to \frac{1}{2r} \frac{r}{r+\mu}$. Given (29), the condition $1/8r < CS((\Delta; \mu^{Crit}(f), r)$ is equivalent to $f < \frac{1}{32r}$. On the other hand, given the constraint $k \ge k_A$, there is a positive investment in data profiling by the monopolist only if $k_A < \frac{3+4rf-2\sqrt{2+8rf}}{4r^2}$. For an equilibrium with strategic consumers to admit positive investment in data collection it is then necessary and sufficient that $f < \frac{1}{32r}$ and $k_A < \frac{3+4rf-2\sqrt{2+8rf}}{4r^2}$. In Figure 7, we picture (for r = 1), the areas in the (f, k) -space where the firm invests in profiling. The colored areas (blue and green) correspond to the couples of values of f and k_A for which the monopolist collects consumers information when consumers are passive, the blue area corresponds to the couples of values of f and k_A for which it invests in profiling when consumers are actively engaged in lobbying for protective data protection laws (strategic customers).

It turns out that consumers' lobbying activity reduces the scope for information collection while not necessarily eliminating it. It can indeed be argued that f being the cost to be incurred simply to collect and to keep the coarsest information (when customers buy the good for the first time), which can be done very simply by hand, f is not likely to



FIGURE 7 The positive profiling investment area with and without strategic customers [Color figure can be viewed at wileyonlinelibrary.com]

be very great. So the most usual effect of protective legislations is probably to reduce the accuracy of the firm's information on its old customers, rather than preventing it from collecting some information.

9 CONCLUSION

In this paper we have analyzed a dynamic model in which the accuracy of the information that a nondurable good monopolist obtains on its previous customers depends on the amount that it invests in profiling, whereas the consumers influence the efficiency of this investment via lobbying over data protection legislation. Given the precision of information the firm obtains, a dynamic game takes place between it and the consumers.

At one extreme, we have the case of absence of learning: while the monopolist knows the distribution of the WTP of his customer base, he does not know the WTP of any particular customer, he cannot tell one consumer from another and does not even know if a given customer has purchased from him before. At the other extreme, we have the case of a monopolist who can discover the customers' exact WTP immediately after they have made their first purchase. In intermediate cases, the monopolist can recognize former customers and remember their first date of purchase, and at any time, the fraction of former customers whose exact WTP remains unknown to him (called the unidentified fraction) is a declining function of the distance between that time and the time of their first purchases. The rate of decline of this fraction is a measure of the precision of the monopolist's information (or, equivalently, the accuracy of his customer profiling technology). If this rate of decline is zero, the monopolist only knows the date of first purchase of each of his former customers. If the rate of decline is infinite, the monopolist discovers each customer's WTP immediately, and is able to extract all their surplus when they return in future periods. If the rate of decline is strictly positive and finite, the monopolists extracts all the surplus of the old customers whose WTP have become known while proposing a vintage-specific price to the other ones.

We have shown that the MPE of this model is such that the profit is non-monotone in the degree of precision of the firm's information. Starting from a scenario of absence of learning, profits suffer a downward jump when the monopolist becomes able to group his former customers according to their date of first purchase to practice price discrimination. We show that there exists a unique threshold level of precision of information such that as long as the ability to learn about customers' WTP is below that threshold, profit remains inferior to the static monopoly profit. Above this threshold, profit increases when the

degree of precision of the information increases, reaching its greatest level when the firm learns instantaneously the WTP of his customers. Interestingly, the aggregate consumers surplus varies exactly in the opposite way. Accordingly, it is not clear that customers' profiling is always good for firms (or bad for consumers).

The outcome of the comparison depends crucially on the accuracy of the technologies used by the firms for consumers' profiling, as well as on the extent of restrictions introduced by consumer protection legislation (which, by limiting the extent of consumer profiling, may shape in important ways the profit, consumer surplus and welfare effects of price discrimination strategies relying on consumers' profiling). We have shown that, to be beneficial, the monopolist's investment in data collection, and the resulting degree of precision of its information, must be large enough. In equilibrium the firm never invests to obtain only a very coarse information on its old customers. In addition, when the customers influence, via lobbying, the shape of the data protection legislation, we have shown that this may lead to no investment in profiling but that, most likely, it is in consumers' interest to assure that the firm is still interested to invest in customer profiling, though less than the firm would do in the presence of inactive consumers, who do not make any effort to influence the firm's ability to engage in customer recognition.

ACKNOWLEDGMENTS

This study was supported by the French National Research Agency Grant ANR-17-EURE-0020, and by the Excellence Initiative of Aix-Marseille University—A*MIDEX. This study has been financed by projects NORTE-01-0145-FEDER-028540 and POCI-01-0145-FEDER-006890.

ORCID

Ngo Van Long https://orcid.org/0000-0001-6211-7229

ENDNOTES

- ¹ As defined in the Cambridge Dictionary, big data is "very large sets of data that are produced by people using the internet, and that can only be stored, understood, and used with the help of special tools and methods."
- ² "Uber Starts Charging What It Thinks You're Willing to Pay," https://www.bloomberg.com/news/articles/2017-05-19/uber-s-future-may-rely-on-predicting-how-much-you-re-willing-to-pay.
- ³ This does not require very sophisticated or very costly means of information collection, storage and treatment and may only need the firm to keep a record by hand.
- ⁴ With the use of tracking tools such as cookies, web beacons, or Etags.
- ⁵ This requires the use of digital technologies and big data.
- ⁶ The polar case where this probability is zero corresponds to the main model analyzed by Laussel et al. (2020a), who specified that the monopolist can never discover a consumer's exact WTP, and thus can only classify customers according to their first date of purchase, that is, customers with different WTPs who make their first purchase on the same date are bunched together. The other polar case obtains under perfect information acquisition: as soon as a customer has made her first purchase, her exact WTP is instantaneously discovered through the analysis of big data.
- ⁷ Although there is inherent randomness concerning the time of discovery of the WTP of any given former customer, in the aggregate, as there is a continuum of customers, one can suppose that on average, the fraction of unidentified customers of any given vintage declines at a constant exponential rate, which in our model will be denoted by μ (please see Section 3 for more details).
- ⁸ In particular, it takes its maximum value when profits take their minimum value.
- ⁹ We are grateful to an anonymous referee for suggesting us this expression.
- ¹⁰ The direct effect on profits (i.e., for given rivals' prices) is always positive while the strategic one is negative due to the increased competition.
- ¹¹ This is the case where the strong market of one firm is the weak market of the other.
- ¹² Bearing in mind that the Hotelling model satisfies best-response asymmetry assumed in Corts (1998), it appears that the robustness of this result is not clear.
- ¹³ Fudenberg and Tirole (2000) considered a two-period model of a Hotelling duopoly. Chen (1996) studied the case of duopoly in the presence of consumers' switching costs. Choe et al. (2018) showed the multiplicity of equilibria under the assumption that the duopolists do not have the same information.
- ¹⁴ See Stokey (1981), Bulow (1982), Gul et al. (1986), Bond and Samuelson (1987), Kahn (1987), Hart and Tirole (1988), Karp (1996), Laussel et al. (2015), and Long (2015), for a survey.

- ¹⁵ In a companion paper, Laussel et al. (2020b) considered the case of a monopolist producing vertically differentiated products. The focus of that paper is on the existence of multiple Markov-perfect equilibria.
- ¹⁶ As a referee points out, there is a parallel between the two types of price discrimination in this paper (one harmful to profit, while the other, based on exact WTP, increases it) and the idea that oligopoly price discrimination is harmful to industry profit when it is based on consumers' differences in cross-firm price elasticity but is beneficial to profit when it is based on consumers' differences in valuation (or price elasticity) for the product. See, for example, Holmes (1989) and Chen (1999). It would also be worthwhile to extend the literature on behavior-based price discrimination to situations where oligopolists can also uncover some consumers' WTP.
- ¹⁷ This investment captures the costs to obtain the hardware, software and human resources (e.g., data analysts) that will be necessary to subsequently engage in price discrimination. The specification of such costs will be presented in more detail in Section 7.
- ¹⁸ We suppose that, conditional on the fact that a given customer's exact WTP has not been uncovered at time *t*, the probability that it will be discovered over the next short time interval dt is μdt , where μ is a positive constant. In other words, μ is the arrival rate of a Poisson process.
- ¹⁹ Notice that both δ and β increase when Δ decrease. They both tend toward 1 when $\Delta \to 0$. When $\Delta \to \infty$, they both tend toward 0, and by L'Hospital's rule, the ratio $\frac{\delta}{\beta} \to \frac{\mu}{r}$. To ascertain the effects of a changing degree of the coarseness of information on the monopolist's profits, we shall have to consider how these profits change when δ changes for a fixed value of β : this amounts to analyze the effect of changes in μ for a fixed length Δ of the period of commitment.
- ²⁰ Since $\theta \in [0, 1]$, we set $\theta_0 = 1$.
- ²¹ The superscript *I* in $p^{I}(\theta)$ refers to identified customers.
- ²² We will interchangeably refer to $U_n(\theta_{n+1})$ as the informational rent of the period *n* marginal new customer.
- ²³ The RHS of the first line (respectively, second line) of Equation (8) is equal to $\beta U_{n+1}(\theta_{n+1})$ in Case A (respectively, Case B).
- ²⁴ We could use a more general conjectured linear-quadratic expectations functions, and show that consistency reduces to the linear rule.
- ²⁵ $V(\Theta(n))$ depends also on $\Theta(n-1)$ and all the other past values of the state variable. This is omitted for the sake of notational simplicity. Note that in period *n* the historical values of the state variable (namely $\Theta(n-1)$, $\Theta(n-2)$, ... $\Theta(0)$), have no relevance for the decision on the optimal $\Theta(n + 1)$. Only $\Theta(n)$ is relevant. Note also that $\prod_{n=1}^{F}$ depends on $\Theta(n)$ and on earlier values.
- ²⁶ This follows from h(.) being increasing in δ so that the equibrium value of γ is increasing in δ . Notice that k(.) is clearly increasing in δ as well, when γ is not too small.
- ²⁷ Notice that this is to be interpreted as following from an underlying relationship between γ and μ for a given value of Δ since $\delta = e^{-\mu\Delta}$.
- ²⁸ The same color code applies thereafter in Figures 2, 4, 5, and 6.
- ²⁹ The green part of the curve corresponds to the values of δ such that $\gamma^e(0.8, \delta) \le 1/2$ (market expansion by large steps). This occurs when δ is small enough. Here, we have used the formula for $\Phi(\beta)$ in Remark 1.
- ³⁰ For instance $\Pi^*(+\infty; r, r) = \Pi^*(0; r, \mu) = \frac{1}{4r}$ but $\Pi^*(\Delta; r, r)$ is first increasing and then decreasing in Δ .
- ³¹ We thank an anonymous referee for suggesting us this term.
- ³² It is equally easy to deal with a quadratic cost function.
- ³³ Straightforwardly $\overline{k}(f) \to 0$ as $f \to \Pi^*(\beta, 0) 1/4r$.
- ³⁴ Indeed when $\mu^* = 0$, the monopolist's net profit when collecting the coarsest information is negative whenever f is positive.
- ³⁵ An alternative interpretation of the model is that, instead of having a continuum of consumers, there is one consumer whose type is a random variable which is realized only after the legislation on data protection is enacted.
- ³⁶ In period *n*, all new customers are offered the same introductory price $p^{u}(n, n)$ and therefore the difference in surpluses gained by type θ and type θ_{n+1} is also $\left(\frac{1-\beta}{r}\right)(\theta \theta_{n+1})$.
- ³⁷ Note that Π_n^F is independent of $\Theta(n + 1)$.
- ³⁸ This analytical result is confirmed when using the ManipulatePlot command in Mathematica to plot the RHS of both equations as functions of γ for all possible values of $(\beta, \delta) \in [0, 1]^2$.
- ³⁹ It obviously follows from Remark 1 that $\gamma^e(e^{-r\Delta}, e^{-\mu\Delta}) \to 1/2$ as $\Delta \to +\infty$.
- ⁴⁰ More precisely this ratio tends toward N/D where $D \equiv (1 \gamma(0))(4\gamma(0) 1)$ and $N \equiv -\mu [2 + (\gamma(0) 4)]\gamma(0) + r(\gamma(0) 1) [2 + \gamma(0)(2\gamma(0) 3)].$

REFERENCES

Acquisti, A., Taylor, C., & Wagman, L. (2016). The economics of privacy. Journal of Economic Literature, 54(2), 442-492.

Acquisti, A., & VArian, H. R. (2005). Conditioning prices on purchase history. Marketing Science, 24(3), 367-381.

Blattberg, R. C., & Deighton, J. (1991). Interactive marketing: Exploiting the age of addressability. Sloan Management Review, 33(1), 5–15.

- Bond, E., & Samuelson, L. (1987). The coase conjecture need not hold for durable good monopolies with depreciation. *Economics Letters*, 24, 93–97.
- Bulow, J. I. (1982). Durable goods monopolists. Journal of Political Economy, 90, 314-332.
- Chen, Y. (1997). Paying customers to switch. Journal of Economics and Management Strategy, 6, 877-897.
- Chen, Y. (1999). Oligopoly price discrimination and resale price maintenance. RAND Journal of Economics, 30(3), 441-455.
- Chen, Y., & Pearcy, J. (2010). Dynamic pricing: When to entice brand switching and when to reward consumer loyalty. *RAND Journal of Economics*, *41*, 674–685.
- Choe, C., King, S., & Matsushima, N. (2018). Pricing with cookies: Behavior-based price discrimination and spatial price competition. *Management Science*, 64, 5669–5687.
- Coase, R. (1972). Durability and monopoly. Journal of Law and Economics, 15, 143-149.
- Corts, K. S. (1998). Third-degree price discrimination in oligopoly: All-out competition and strategic commitment. *The RAND Journal of Economics*, 29, 306–323.
- Esteves, R. B. (2009). Price discrimination with partial information: Does it pay-off? Economics Letters, 105, 28-31.
- Esteves, R. B. (2010). Pricing with customer recognition. International Journal of Industrial Organization, 28, 669-681.
- Fudenberg, D., & Tirole, J. (2000). Customer poaching and brand switching. RAND Journal of Economics, 31, 634-657.
- Fudenberg, D., & Villas-Boas, J. M. (2006). Behavior-based price discrimination and customer recognition. In T. Hendershot (Ed.), *Economics and information systems*. Elsevier.
- Fudenberg, D., & Villas-Boas, J. M. (2012). Price discrimination in the digital economy. In M. Peitz & J. Waldfogel (Eds.), Oxford handbook of the digital economy. Oxford University Press.
- Gehrig, T., Shy, O., & Stenbacka, R. (2011). History-based price discrimination and entry in markets with switching costs. *European Economic Review*, 55, 732–739.
- Gehrig, T., Shy, O., & Stenbacka, R. (2012). A welfare evaluation of history-based price discrimination. *Journal of Industry, Competition and Trade*, *12*, 373–393.
- Gul, F., Sonnenschein, H., & Wilson, R. (1986). Foundations of dynamic monopoly and the coase conjecture. *Journal of Economic Theory*, 39, 155–190.
- Hart, O., & Tirole, J. (1988). Contract renegotiation and coasian dynamics. Review of Economic Studies, 55, 509-540.
- Holmes, T. J. (1989). The effects of third-degree price discrimination in oligopoly. American Economic Review, 79(1), 244-250.
- Kahn, C. (1987). The durable goods monopolist and consistency with increasing costs. *Econometrica*, 54, 274–294.
- Karp, L. (1996). Depreciation erodes the coase conjecture. European Economic Review, 40, 473-490.
- Laussel, D., Long, N. V., & Resende, J. (2015). Network effects, aftermarkets and the coase conjecture. *International Journal of Industrial Organization*, *41*, 84–96.
- Laussel, D., Long, N. V., & Resende, J. (2020a, Fall). The curse of knowledge: Having access to customer information can reduce monopoly profits. *The RAND Journal of Economics*, 51(3), 650–675.
- Laussel, D., Long, N. V., & Resende, J. (2020b, May). Quality and price personalization under customer recognition: A dynamic monopoly model. Journal of Economic Dynamics and Control, 114, 1–25.
- Liu, Q., & Serfes, K. (2004). Quality of information and oligopolistic price discrimination. *Journal of Economics & Management Strategy*, 13, 671–702.
- Long, N. V. (2015). Dynamic games between firms and infinitely-lived consumers: A review of the literature. *Dynamic Games and Applications*, 5, 467–492.
- Stokey, N. L. (1981). Rational expectations and durable goods pricing. Bell Journal of Economics, 12, 112-28.
- Stole, L. A. (2007). Price discrimination and competition. In M. Armstrong & R. H. Portes (Eds.), *Handbook of industrial organization* (Vol. 3, pp. 2221–2299). North-Holland.
- Tirole, J. (1988). The theory of industrial organization. MIT Press.
- Thisse, J. F., & Vives, X. (1988). On the strategic choice of spatial price policy. American Economic Review, 78, 122-137.
- Villas-Boas, M. (1999). Dynamic competition with customer recognition. RAND Journal of Economics, 30, 604-631.

APPENDIX A

Proof of the participation constraint (8). Consider a non-marginal new customer in period *n*, that is, a customer whose type is θ , where $\theta_n \ge \theta > \theta_{n+1}$. What is the difference between her expected intertemporal utility and that of the marginal customer θ_{n+1} ? Let us denote this difference by

$$D_n(\theta, \theta_{n+1}) \equiv U_n(\theta) - U_n(\theta_{n+1}).$$
(A1)

To calculate this difference, let us recall that, for any integer j > n, the number δ^{j-n} is the probability of the event that the WTP of a period *n* first-time customer remains undiscovered at the beginning of period *j*. There are two cases to be considered: Case *A* (characterized by $\theta_{n+1} \ge (1/2)\theta_n$) and Case *B* (characterized by $\theta_{n+1} \le (1/2)\theta_n$). Case *A* (respectively, *B*) means that the market expansion in period *n* is small (resp., big).

From Equation (2), we know that in Case *A* the price charged (in any future period j > n) to unidentified former customers of vintage *n* is $p^{u*}(n, j) = \theta_{n+1}$ and consequently, compared with type θ_{n+1} , all those with WTP $\theta > \theta_{n+1}$ enjoy an excess surplus of $\left(\frac{1-\beta}{r}\right)(\theta - \theta_{n+1})$ per period.³⁶ Summing across periods (including period *n* when all new customers are charged a common introductory price) with proper accounting for time and for the probability of becoming identified in the future, we obtain for Case *A*,

$$D_n(\theta, \theta_{n+1}) = \left(\frac{1-\beta}{r}\right)(\theta - \theta_n)(1 + \delta\beta + \delta^2\beta^2 + \delta^3\beta^3 + \cdots)$$
(A2)

$$=\frac{(1-\beta)(\theta-\theta_n)}{r(1-\delta\beta)}.$$
(A3)

In Case *B*, the price offered in any future period j > n to unidentified vintage *n* customers is $p^{u*}(n, j) = (1/2)\theta_n > \theta_{n+1}$, and consequently, whereas in the period *n* the excess surplus of type θ over type θ_{n+1} is $\left(\frac{1-\beta}{r}\right)(\theta - \theta_n)$ as they are offered an attractive introductory price $p^u(n, n) < \theta_{n+1}$, in later periods j > n, the excess surplus is only $\left(\frac{1-\beta}{r}\right)\max\left\{\theta - \frac{\theta_n}{2}, \theta\right\}$. Summing across periods, we get for Case *B*,

$$D_{n}(\theta, \theta_{n+1}) = \left(\frac{1-\beta}{r}\right)(\theta - \theta_{n}) + \left[\left(\frac{1-\beta}{r}\right)\max\left\{\theta - \frac{\theta_{n}}{2}, \theta\right\}\right]\sum_{k=1}^{\infty} (\delta\beta)^{k} = \left(\frac{1-\beta}{r}\right)\left[\theta - \theta_{n} + \frac{\delta\beta}{1-\delta\beta}\max\left\{\theta - \frac{\theta_{n}}{2}, \theta\right\}\right].$$
(A4)

Using Equations (A3) and (A4), and the definition (A1), we obtain the life-time net utility, $U_n(\theta)$, of a type θ customer (where $\theta \ge \theta_{n+1}$) who makes her first purchase in period *n*, as equal to the life-time utility of the marginal period *n* new customer, plus an expected utility differential between the two types over an infinite horizon. The life-time net utility $U_{n+1}(\theta)$ of any type θ who deviates to buy for the first time in period n + 1 can be defined in a similar way.

Now
$$U_{n+1}(\theta_{n+1}) = \beta(U_{n+1}(\theta_{n+2}) + D_{n+1}(\theta_{n+1}, \theta_{n+2}))$$
 implies (8).

Proof of Lemma 1. Given the monopolist's cut-off parameter γ , using our conjectured expectations rule (11), the rational expectations requirement, with the help of Equation (8), can be rewritten as

$$\lambda \theta_{n} = \beta \left[\frac{(1-\beta)(\theta_{n+1}-\theta_{n+2})}{r(1-\delta\beta)} + \lambda \theta_{n+1} \right] \text{ if } \theta_{n+2} \ge \frac{\theta_{n+1}}{2},$$

$$\lambda \theta_{n} = \beta \left[\frac{1-\beta}{r} \left(\theta_{n+1}-\theta_{n+2} + \frac{(\delta\beta)\theta_{n+1}}{2(1-\delta\beta)} \right) + \lambda \theta_{n+1} \right] \text{ if } \theta_{n+2} \le \frac{\theta_{n+1}}{2}.$$
(A5)

Using the conjectured equilibrium cut-off rule, we have $\Theta(n + 1) = \gamma \Theta(n)$ and $\Theta(n + 2) = \gamma^2 \Theta(n)$, then Equation (A5) becomes

$$\begin{split} \lambda \theta_n &= \beta \left[\frac{1 - \beta}{r(1 - \delta\beta)} (\gamma \theta_n - \gamma^2 \theta_n) + \lambda \gamma \theta_n \right] \text{ if } \gamma \geq \frac{1}{2}, \\ \lambda \theta_n &= \beta \left[\frac{1 - \beta}{r} \left(\gamma \theta_n - \gamma^2 \theta_n + \frac{(\delta\beta) \gamma \theta_n}{2(1 - \delta\beta)} \right) + \lambda \gamma \theta_n \right] \text{ if } \gamma \leq \frac{1}{2}. \end{split}$$

from which we obtain the "best response" of consumers expectations to the monopolist's γ :

$$r\lambda = \frac{\beta\gamma(1-\beta)(1-\gamma)}{(1-\beta\gamma)(1-\beta\delta)} \text{ if } 1 \ge \gamma \ge \frac{1}{2},$$

$$r\lambda = \frac{\beta\gamma(1-\beta)\left(1-\gamma+\frac{\delta\beta}{2(1-\delta\beta)}\right)}{(1-\beta\gamma)} \text{ if } \gamma \le \frac{1}{2}.$$
(A6)

For later use, it is convenient to define the function $\Lambda(\gamma, \beta, \delta)$ by

$$\Lambda(\gamma,\beta,\delta) \equiv \frac{\beta\gamma(1-\beta)(1-\gamma)}{(1-\beta\gamma)(1-\beta\delta)}.$$
(A7)

Then the consumers' best reply λ to the monopolist's cut off choice γ can be expressed as follows:

$$r\lambda = \Lambda(\gamma, \beta, r) \quad \text{if} \quad 1 \ge \gamma \ge \frac{1}{2},$$

$$r\lambda = \Lambda(\gamma, \beta, r) - \frac{\beta^2 (1 - \beta)\gamma\delta(1 - 2\gamma)}{2(1 - \beta\gamma)(1 - \beta\delta)} \quad \text{if} \quad \gamma \le \frac{1}{2}.$$
(A8)

Proof of Lemma 2. Maximizing the right-hand side of the Bellman Equation (13) with respect to $\Theta(n + 1)$, one obtains the first-order condition³⁷

$$(1-\beta)(\Theta(n) - 2\Theta(n+1)) + r\lambda\Theta(n) + r\beta V'(\Theta(n+1)) = 0,$$
(A9)

where V' denotes the derivative of the value function.

From Equation (13), we obtain

$$V(\Theta(n+1)) = \max_{\Theta(n+2)} \left\{ \Pi_{n+1}^F + \Pi_{n+1}^N(\Theta(n+1), \Theta(n+2), \lambda) + \beta V(\Theta(n+2)) \right\}.$$
 (A10)

Differentiating Equation (A10) with respect to $\Theta(n + 1)$, we get

$$V'(\Theta(n+1)) = \frac{\partial \Pi_{n+1}^F}{\partial \Theta(n+1)} + \frac{\partial \Pi_{n+1}^N}{\partial \Theta(n+1)},$$

where

$$\frac{\partial \Pi_{n+1}^{F}}{\partial \Theta(n+1)} = \frac{\partial \pi^{*}(n, n+1)}{\partial \Theta(n+1)}$$
$$= \begin{cases} -\frac{1-\beta}{r} [(1-\delta)\theta_{n+1} + \delta(2\theta_{n+1} - \theta_{n})] & \text{if } \theta_{n+1} \ge \frac{\theta_{n}}{2} \\ -\frac{1-\beta}{r} (1-\delta)\theta_{n+1} & \text{if } \theta_{n+1} \le \frac{\theta_{n}}{2} \end{cases}$$

and, from $\Pi_{n+1}^N = \left[\theta_{n+1} - \theta_{n+2}\right] \left[\frac{1-\beta}{r}\theta_{n+2} - \lambda\theta_{n+1}\right],$

$$\frac{\partial \Pi_{n+1}^{N}}{\partial \Theta(n+1)} = \left(\frac{1-\beta}{r}\right) \theta_{n+2} - \lambda \theta_{n+1} - \lambda \left[\theta_{n+1} - \theta_{n+2}\right].$$

Thus we obtain

$$V'(\Theta(n+1)) = -\lambda [2\Theta(n+1) - \Theta(n+2)] + \frac{1-\beta}{r} [\Theta(n+2) - (1-\delta)\Theta(n+1) + \delta(\Theta(n) - 2\Theta(n+1))].$$
(A11)

if $\Theta(n+1) \ge \frac{1}{2}\Theta(n)$ and

$$V'(\Theta(n+1)) = -\lambda [2\Theta(n+1) - \Theta(n+2)] + \frac{1-\beta}{r} [\Theta(n+2) - (1-\delta)\Theta(n+1)].$$
(A12)

if $\Theta(n+1) \leq \frac{1}{2}\Theta(n)$.

Notice that $V'(\Theta(n + 1))$ is continuous at $\Theta(n + 1) = \frac{1}{2}\Theta(n)$. Moreover, the RHS of (A11) takes on a smaller value than the RHS of (A12) when $\Theta(n + 1) \ge \frac{1}{2}\Theta(n)$, and conversely, when $\Theta(n + 1) \le \frac{1}{2}\Theta(n)$, the RHS of (A12) takes on a smaller value than the RHS of (A11). This allows us to express $r\beta V'(\Theta(n + 1))$ as

$$-r\lambda\beta[2\Theta(n+1) - \Theta(n+2)] +\beta(1-\beta)[\Theta(n+2) - (1-\delta)\Theta(n+1)] +\beta(1-\beta)\delta\min\{0,\Theta(n) - 2\Theta(n+1)\}.$$
(A13)

Substituting the above expressions for $r\beta V'(\Theta(n + 1))$ into the first-order condition (A9), we obtain the Euler equation

$$0 = (1 - \beta)[(\Theta_n - 2\Theta_{n+1}) + \beta(\Theta_{n+2} - (1 - \delta)\Theta_{n+1})] + r\lambda\Theta_n + \beta\delta(1 - \beta)\min\{0, \Theta_n - 2\Theta_{n+1}\} - r\lambda\beta[2\Theta_{n+1} - \Theta_{n+2}].$$
(A14)

It is convenient to define the following functions H(.) and K(.) (which apply for the case $\Theta_n - 2\Theta_{n+1} \le 0$ and $\Theta_n - 2\Theta_{n+1} \ge 0$, respectively):

$$H(.) \equiv (1 - \beta)[(\Theta_n - 2\Theta_{n+1}) + \beta(\Theta_{n+2} - (1 - \delta)\Theta_{n+1})] + r\lambda\Theta_n -r\lambda\beta[2\Theta_{n+1} - \Theta_{n+2}] + \beta\delta(1 - \beta)[\Theta_n - 2\Theta_{n+1}]$$

and

$$K(.) \equiv H(.) - \beta \delta (1 - \beta) [\Theta_n - 2\Theta_{n+1}].$$

Then the Euler equation may be written as

$$0 = \min\{H(\Theta_n, \Theta_{n+1}, \Theta_{n+2}), K(\Theta_n, \Theta_{n+1}, \Theta_{n+2})\}.$$
(A15)

That is, the function H(.) applies when $\Theta_{n+1} \ge (1/2)\Theta_n$ (the market is expanded by small steps) and the function K(.) applies when $\Theta_{n+1} \le (1/2)\Theta_n$ (the market is expanded by large steps).

Under the linear cut off rule, the function H(.) applies iff $\gamma \ge 1/2$, in which case we have, from eq. (A6), the following value of $r\lambda$:

$$r\lambda = \Lambda(\gamma, \beta, r) = rac{\beta\gamma(1-\beta)(1-\gamma)}{(1-\beta\gamma)(1-\beta\delta)}.$$

Substituting this $r\lambda$ into Equation (A15), and replacing $\Theta_{n+1} = \gamma \Theta_n$ and $\Theta_{n+2} = \gamma^2 \Theta_n$, we can see that when H(.) applies, we have

$$h(\gamma, \beta, \delta) = (1 - \beta)\{(1 - 2\gamma) + \beta\gamma[\gamma - (1 - \delta)] + \beta\delta(1 - 2\gamma)\} + (1 + \beta\gamma(2 - \gamma))\Lambda(.).$$
(A16)

Similarly, the function K(.) applies iff $\gamma \leq 1/2$, in which case we have, from Equation (A6) the following value of $r\lambda$:

$$r\lambda = \Lambda(\gamma, \beta, \delta) - \frac{\beta^2 (1 - \beta)\gamma \delta (1 - 2\gamma)}{2(1 - \beta\gamma)(1 - \beta\delta)}$$

Substituting this $r\lambda$ into Equation (A15), and using $\Theta_{n+1} = \gamma \Theta_n$ and $\Theta_{n+2} = \gamma^2 \Theta_n$, we we can see that when *K*(.) applies, we have

$$k(\gamma,\beta,\delta) = h(\gamma,\beta,\delta) - (1+\beta\gamma(2-\gamma)) \left[\frac{\beta^2(1-\beta)\gamma\delta(1-2\gamma)}{2(1-\beta\gamma)(1-\beta\delta)} \right] - \beta\delta(1-\beta)[1-2\gamma].$$
(A17)

Proof of Proposition 1. Lemma 2 tells us that given (β, δ) , k(.) as a function of γ applies in the range $0 \le \gamma \le 1/2$, whereas h(.) as a function of γ applies in the range $1/2 \le \gamma \le 1$. Comparing (A16) with (A17) we see that $k(\gamma, \beta, \delta) \ge h(\gamma, \beta, \delta)$ if $\gamma \in [1/2, 1]$ and $k(\gamma, \beta, \delta) \le h(\gamma, \beta, \delta)$ if $\gamma \in [0, 1/2]$, with $k(1/2,\beta,\delta) = h(1/2,\beta,\delta)^{.38}$ Furthermore, for all $(\beta,\delta) \in (0,1) \times (0,1)$, we have $h(\gamma,\beta,\delta) > k(\gamma,\beta,\delta)$ if $\gamma \in [0, 1/2)$ and $h(\gamma, \beta, \delta) < k(\gamma, \beta, \delta)$ if $\gamma \in (1/2, 1]$. This shows that, for any MPE in which the monopolist's optimal cut-off rule and consumers' expectation rules are mutual best-replies, either we have (i) $h(\gamma, \beta, \delta) = 0$ where $0 \le \gamma \le 1/2$, or we have (ii) $k(\gamma, \beta, \delta) = 0$ where $1/2 \le \gamma \le 1$, but not both.

Substituting $r\lambda = \Lambda(\gamma, \beta, r) = \frac{\beta\gamma(1-\beta)(1-\gamma)}{(1-\beta\gamma)(1-\beta\delta)}$ into eq. (A16), we get the following fourth-order polynomial in γ

$$h(\gamma, \beta, \delta) = 1 - \beta^2 \delta^2 - \gamma^4 \beta^2 + \gamma^3 \beta^2 (2 + \beta \delta) + \gamma^2 \beta (2 - \beta - 2\beta \delta - \beta^2 \delta - \beta^2 \delta^2) + \gamma (-2 - \beta + \beta \delta + \beta^2 \delta + \beta^2 \delta^2 + \beta^3 \delta^2) = 0.$$
(A18)

Then, given any pair (β, δ) where $0 \le \beta \le 1$ and $0 \le \delta \le 1$, any real root $\gamma^* \in [1/2, 1]$ of (A18) is the monopolist's equilibrium cut off rule, and the consumers best response to γ^* is given by $r\lambda^* = \Lambda(\gamma^*, \beta, \delta)$. Similarly, iff $\gamma \leq 1/2$, we have, from Equation (A6):

$$r\lambda = \Lambda(\gamma, \beta, \delta) - \frac{\beta^2(1-\beta)\gamma\delta(1-2\gamma)}{2(1-\beta\gamma)(1-\beta\delta)}.$$

Substituting this $r\lambda$ into Equation (A17), we obtain the following fourth-order polynomial in γ :

$$k(\gamma, \beta, \delta) = \frac{1}{2} [2 - 2\beta\delta - \gamma^{4}\beta^{2}(2 - 2\beta\delta) + \gamma^{3}\beta^{2}(4 - 3\beta\delta) -\gamma^{2}(-4\beta + 2\beta^{2} + 6\beta^{2}\delta - 2\beta^{3}\delta^{2}) -\gamma(4 + 2\beta - 6\beta\delta - 3\beta^{2}\delta + 2\beta^{2}\delta^{2})] = 0.$$
 (A19)

Then, given any pair (β, δ) where $0 \le \beta \le 1$ and $0 \le \delta \le 1$, any real root $\gamma^{**} \in [0, 1/2]$ of (A19) is the monopolist's equilibrium cut off rule, and the consumers best response to γ^{**} is given by

$$r\lambda^{**} = \Lambda(\gamma^{**}, \beta, \delta) - \frac{\beta^2(1-\beta)\gamma^{**}\delta(1-2\gamma^{**})}{2(1-\beta\gamma^{**})(1-\beta\delta)}.$$

Proof of Remark 1. We can use the equation $h(1/2, \beta, \delta) = 0$ to characterize the subset *S* of points (β, δ) in the unit square $[0, 1] \times [0, 1]$, such that $\gamma(\beta, \delta) = 1/2$. It is easy to verify that the set *S* can be represented by a curve $\delta = \Phi(\beta)$, where (i) $\Phi(\beta) \equiv \frac{4 - \beta^2 - \sqrt{16(1 - \beta^2) + \beta^2(4 + \beta)}}{4\beta(2 - \beta)}$ for $\beta \in (0, 1]$, (ii) $\Phi'(\beta) > 0$, (iii) $\lim_{\beta \to 1} \Phi(\beta) = (3 - \sqrt{5})/4 \equiv \overline{\delta} < 1$, and (using L'Hospital's rule) $\lim_{\beta \to 0} \Phi(\beta) = 0$. It follows that $\gamma(\beta, \delta) < 1/2$ (i.e., the market expands in large steps) for all points (β, δ) that lie below the curve, and $\gamma(\beta, \delta) > 1/2$ for all points (β, δ) that lie above the curve.³⁹

Proof of Remark 2. In the limiting case when $\Delta \to 0$, we have $\delta \to 1$ and $\beta \to 1$; thus, by continuity, it follows that $\delta > \Phi(1) \ge \Phi(\beta)$ if Δ is small enough. We conclude that, for any given pair $(r, \mu) > (0, 0)$, we have $\gamma^e(r, \mu, \Delta) \ge 1/2$ (the market expands in small steps) for sufficiently small values of Δ .

Proof of Claim 1. The monopolist's equilibrium cut off strategy implies a market expansion rule: by the end of period *n*, where n = 0, 1, 2, ..., the fraction of the customer base that has not been served is $(\gamma^e(\beta, \delta))^{n+1}$. The measure of customers that have been served by the end of period *n* is $(1 - (\gamma^e)^{n+1})$. Given *r* and μ , we consider β and δ as functions of Δ , hence we can define the function $\gamma(\Delta) = \gamma^e(\beta(\Delta), \delta(\Delta))$. Let $t = n\Delta$. Then $(\gamma^e(\beta, \delta))^{n+1} = (\gamma(\Delta))^{\frac{t}{\Delta}+1})$. Since $\lim_{\Delta \to 0} \gamma(\Delta) = 1$ (using Equation A18), we have $\lim_{\Delta \to 0} (\gamma(\Delta)^{\frac{t}{\Delta}+1}) = \lim_{\Delta \to 0} (\gamma(\Delta)^{\frac{t}{\Delta}})$. Notice that $\ln \gamma(\Delta)^{\frac{t}{\Delta}} = \frac{t}{\Delta} \ln \gamma(\Delta)$. To determine the limit value of $\frac{\ln \gamma(\Delta)}{\Delta}$ when $\Delta \to 0$, we have to use L'Hospital's rule and evaluate the ratio of the derivatives of the numerator and the denominator at $\Delta = 0$. This ratio turns out to be equal to $\gamma'(0)$. It follows that $\lim_{\Delta \to 0} (\gamma(\Delta)^{\frac{t}{\Delta}}) = e^{\gamma'(0)t}$.

Now let us determine $\gamma'(0)$. Write the Euler Equation (A18) as

$$E(\Delta, \gamma(\Delta)) = 0.$$

Differentiating this equation (an identity in Δ) with respect to Δ, we obtain $E_{\Delta}^{'}(\Delta,\gamma(\Delta)) + E_{\gamma}^{'}(\Delta,\gamma(\Delta))\gamma'(\Delta) = 0. \text{ Remembering that } \gamma(0) = 1, \text{ it turns}$ $-\frac{E_{\Delta}^{'}(\Delta,\gamma(\Delta))}{E_{\gamma}^{'}(\Delta,\gamma(\Delta))} \to -\infty \text{ as } \Delta \to 0.^{40} \text{ It follows that } \lim_{\Delta \to 0} (\gamma(\Delta)^{\frac{t}{\Delta}+1}) = 0.$ out that the ratio

Proof of Claim 2. From Remark 2, we only need to consider the value of profits as given by (17) (where, of course, $\beta = e^{-r\Delta}$ and $\delta = e^{-\mu\Delta}$). Since $\lim_{\Delta \to 0} \Pi^*(\Delta; r, \mu)$ is undefined, we have to use L'Hospital's rule and evaluate the ratio of the derivatives of the numerator and the denominator at $\Delta = 0$. This ratio equals

$$\frac{\gamma'(0)[-r(r+\mu)+\mu\gamma'(0)]}{r(r+\mu)(r-2\gamma'(0))(r-\gamma'(0))}$$

which itself equals $\frac{\mu}{2r(r+\mu)}$ since, from Claim 1, $\gamma'(0) = -\infty$.